The Collateral Channel: Real Estate Prices and Firm Leverage^{*}

Sevcan Yeşiltaş[†] Department of Economics Johns Hopkins University

JOB MARKET PAPER

November 25, 2015

Please check https://www.dropbox.com/s/7t39ci8zrtnfs4u/jmp_SY.pdf?dl=0 for the latest version.

ABSTRACT: Through the collateral channel, shocks to the value of real estate can have a significant impact on the firms' borrowing capacity. In this paper, I provide evidence on this mechanism by using LTV ratio caps on mortgages in a number of European countries as policy shocks that affect real estate prices. I conduct a difference-in-difference exercise using a unique and comprehensive micro panel data covering both large firms and SMEs. This allows me to better identify and quantify the effects of policy shocks to the value of firm collateral by distinguishing them from local demand shocks and local general equilibrium effects. I find a significant collateral damage on firms' balance sheets, a consequence of LTV policy shock, which in turn caused i) secured debt to decrease in firms with high collateral value more than in firms with low collateral value. These findings document a new evidence on how firms adjust to shocks to the value of collateral through trade credit use. These findings also highlight that macroprudential policies in one sector–such as LTV ratio caps targeting household sector–might result in an unintentional consequence in another sector–such as collateral damage in corporate sector. This is an important caveat that policy makers should consider when implementing macroprudential policy.

JEL-Codes: D22, E58, G21, G28, G30, G32, R30

Keywords: Firm Leverage, Collateral, Real Estate Bubbles, Systemic Risk, Loan-to-Value (LTV) ratios, Macroprudential Policy

^{*}I wish to thank Professor Olivier Jeanne, Şebnem Kalemli-Özcan, and Anton Korinek for invaluable advice, support and guidance. I gratefully acknowledge helpful comments from seminar participants at Johns Hopkins University. All errors are my own.

[†]syesilt1@jhu.edu

1 Introduction

This paper investigates how firms' capital structure decisions respond to changes in collateral value, caused by real estate price shock. The contributions of this paper are twofold. First, using a unique and comprehensive firm-level data, I identify and quantify the impact of a change in the value of real estate assets on firms' debt financing decisions. Second, I provide new evidence on how firms' borrowing capacity, strongly associated with collateral pledging, determines firms' choice between secured and unsecured debt financing. Therefore, this paper assesses the role of collateral pledging in transmitting boom-bust cycles in real estate markets to the corporate sector.

Real estate booms have been often associated with economic and financial busts. As a consequence, academics and policy makers have been trying to understand how these booms are transmitted to the real economy. The relevant theoretical literature suggests that the "collateral channel" might have an important role in transmitting shocks in real estate markets to the real economy: the bursting of a real estate market bubble adversely affects the value of collaterizable real estate assets. Declining collateral values lead to higher cost of external financing which forces firms to decrease borrowing and lower investment leading to a decline in output (Kiyotaki and Moore (1997)). Although there is a significant body of theoretical literature that suggests the significance of the collateral channel, there has been a limited number of empirical studies that identifies and quantifies its economic impact. This paper is one of the papers that attempts to fill this gap in the literature.

In recent years, a number of European countries have experienced a huge increase in real estate prices associated with rapid credit growth and lax lending standards. They applied maximum loan-to-value ratios on mortgages (henceforth referred to as "LTV ratio caps") with the aim of damping credit growth and price inflation in housing markets.¹ Their policy experiment provides an ideal setting: it solves the endogeneity problem typically encountered in this type of study. Tightening of LTV ratio of mortgages led to a slowdown in price inflation in housing markets, a policy shock that is plausibly exogeneous to any individual firm. This policy shock has a general equilibrium effect through the demand and supply of houses on collateral values, and it is affecting firm financing decisions through collateral channel.

My hypothesis is that if LTV ratio cap is effective in curbing borrowers' demand by tightening borrowers' capacity to borrow, this will have a negative effect on real estate prices, then firms who had higher valued collateral pre-LTV policy shock should experience a bigger drop after-LTV policy shock in their secured borrowing relative to firms who had lower valued collateral pre-LTV policy shock. In order to test this hypothesis, I develop a difference-in-difference estimation with firm fixed effects. The estimation, by interacting firm collateral with a time dummy that separates the period before and after the LTV policy shock, captures before-after shock difference in secured borrowing of firms with similar collateral on their balance sheet prior the LTV policy shock. The inclusion of firm fixed effects is important because the LTV policy shock might affect different firms

¹LTV ratio cap is a cap on the ratio of the value of the loan (L) relative to the value of the underlying collateral (V). LTV ratio caps impose a limit on borrowers' capacity to borrow on collateralized lending. LTV ratio caps are not harmonised under the Capital Requirements Directive (CRD)/Capital Requirements Regulation (CRR), but rather are implemented at national discretion. Therefore, LTV ratio caps can be viewed as a recommendation or restriction of credit standards that banks should follow when issuing loans. I provide further details about this policy in section 2.

differently due to unobserved firm characteristics. For example, if high risk taker firms hold less collateral on average, such firms might be affected differentially from the LTV policy shock. In the estimation, this average affect will be fully absorbed by firm fixed effects, and will not invalidate the identification. The identification will come from the timing of the policy shock interacted with predetermined value of firm collateral, which is not allowed to move with the policy shock.

Using a large sample containing non-financial non-real estate firms, I study the episodes of LTV ratio caps on mortgages in Europe (Bulgaria: 2004–2007, Hungary: 2010-present, The Netherlands: 2008-present, Norway: 2010-present, Sweden: 2010-present, and Turkey: 2010-present). In benchmark difference-in-difference estimation, I regress firm leverage (total debt to total assets) on its determinants where collateral is interacted with LTV dummy. Since this is a dummy that takes value 1 in the year(s) when the LTV ratio cap is place, it allows me to understand how the impact of collateral on firm leverage is affected by this policy shock through its effect on real estate prices. I find a significant collateral damage effect on firm leverage after the LTV ratio cap. This effect is different for firms with differential collateral values: the LTV ratio cap decreased leverage in firms with high collateral value by 0.9 percentage point more than in firms with low collateral value. The result is robust to different specifications.

This paper is related to the recent empirical studies by Gan (2007), Chaney, Sraer, and Thesmar (2012) and Cvijanović (2014) who used local variations in real estate price movements as shocks to the value of collateral to identify the causal impact of financial shocks on firms' decisions. Gan (2007) investigated Japan's land market collapse in early 1990's. Within a difference-in-difference like approach, she estimated pre-shock landholdings in 1989 as an exogeneous instrument to identify the effect of the bursting of real estate bubble on the average investment rate during five years after the shock, 1994 to 1998. She showed that land-holding Japanese firms were more affected by the shock than firms with no land. Chaney, Sraer, and Thesmar (2012) and Cvijanović (2014) focused on US real estate price boom between 1993–2006 that resulted in a large dispersion in real estate price movements between US states and cities. They followed an instrumental variables approach to isolate the variation in local real estate prices, which may be endogeneous to firms' decisions. They both showed that firms significantly change their decisions in response to collateral value appreciation.

My relative contribution is twofold. First, the difference-in-difference approach and the richness of data set I use allow me to better identify the effects of shocks to collateral values by distinguishing them from local demand shocks and local general equilibrium effects. One concern with the existing estimates is that the correlation between changes in collateral values and changes in firms' decisions might be observed due to a reverse causality problem: say, in response to increases in investment (accompanied by increases in borrowing), large publicly traded firms may have a significant impact on local real estate prices through increases in local business activity and demand for labor. Indeed, in the existing papers, focusing only on large publicly traded firms might be biasing the estimates. However, by using a comprehensive sample dominated by small firms, I minimize the possibility of such reverse causality problem in the estimation.

There is another concern with existing estimates that unobserved variation within a particular location and year might drive the results. To be precise, consider the possibility that real estate price shocks are actually affecting the balance sheet of consumers, not of firms, and this might drive the results through the changes in local demand. I solve this by using four-digit sector-year fixed effects to control for demand. These effects will absorb the impact of changes in local demand for the four-digit sector that the firms operate in. I assume that most of the changes in local demand derive from narrowly defined sector-specific factors. The identifying assumption requires that firms with high collateral value are subject to similar local demand shocks as firms with low collateral value in the same four-digit sector and any remaining variation in firm specific demand conditions does not vary systematically by the collateral value. I am not the first to control for demand using sector fixed effects (e.g., Nanda and Nicholas (2014) and Acharya, Eisert, Eufinger, and Hirsch (2014)) but to the best of my knowledge, Kalemli-Ozcan, Laeven, and Moreno (2015) and this paper are the first to allow these effects to vary at a very fine level of sector classification.

Second, existing papers use only large publicly traded firms as in most of the papers studying investment and capital structure decisions in the literature. Being less financially constrained, such firms are least likely to pledge collateral when they borrow from financial institutions. This might lead to a downward bias of the effect of real estate price shock. However, I develop a unique and comprehensive data set that covers not only large publicly traded firms, but also small and medium private firms. The inclusion of small and medium firms is crucial given the structure of European economies. Europe consists of bank-dominant economies tilted toward externally dependent small and medium enterprises (SMEs), and among all sources of external financing, European firms typically prefer debt to finance working capital and/or investment (e.g., Kalemli-Ozcan, Laeven, and Moreno (2015)).^{2,3} Tangible assets have been very often pledged as collateral in business lending, and European banks heavily prefer real-estate as collateral especially for SMEs.^{4,5} Therefore, with the inclusion of SMEs, I can obtain more accurate estimates of collateral in the analysis of firms' debt financing decisions, then precisely link them to the actual changes in aggregate corporate debt movements.

Turning to the results, I find a significant collateral damage effect on firm leverage both at the micro level and at the aggregate level. The micro estimates I obtain from difference-in-difference estimation show collateral damage can explain around 16 % of the actual decline in aggregate corporate sector. Aggregate corporate debt patterns might mask how financing patterns of SMEs respond to real estate price shocks. However, financing patterns studied in this setting provide important insights: tightening of LTV ratios result in a significant collateral damage in firms'

 $^{^{2}}$ According to the recent report of European Commission, across the EU-28 in 2013, some 21.6 million SMEs (firms with less than 250 employees) in the non-financial corporate sector employed 88.8 million people and generated 3,666 trillion euro in terms of value added. In other words, 99 out of every 100 businesses are SMEs, as are 2 in every 3 employees and 58 cents in every euro of value added.

 $^{^{3}}$ Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015) shows how well the data used in this paper covers the universe of European firms compared to official statistics from Eurostat along several dimensions. See their paper for further statistics.

⁴One of the recent reports developed by International Finance Corporation states that while land and buildings are widely accepted as collateral for loans, the use of movable collateral (such as inventory, accounts receivables, crops, machinery and equipment) is restricted due to lack of functioning laws and registries to govern secured transactions. For further details, see Secured Transactions Systems & Collateral Registries Toolkit (2010).

⁵For instance, according to the World Bank Investment Climate Survey of 6,511 firms in 24 European countries, nearly 63% of the loans required collateral, and 77% of these loans are secured by real estate (land, buildings, houses owned by the entrepreneurs). For further details, see http://www.enterprisesurveys.org

balance sheets by lowering real estate prices. This in turn causes a bigger drop in secured lending of SMEs relative to large firms with similar collateral on their balance sheet prior the LTV policy shock.

The collateral channel also has important implications for the usage of trade credit.⁶ According to the "balance-sheet channel," changes in monetary policy have potential impact on firms' ability to borrow by changing the value of collateral (Bernanke and Gertler (1995)). The empirical literature has found that trade credit usage changes as a response to monetary policy shocks and business downturns (e.g., Choi and Kim (2005), Mateut, Bougheas, and Mizen (2006), and Nilsen (2002)). To the best of my knowledge, this paper is the first to provide evidence on how the usage of trade credit responds to changes in collateral value, caused by real estate price shock. In this paper, I ask this specific question: Did firms that experience a collateral damage after the LTV ratio cap turn to trade credit as an alternative source of finance?

The sample used in this paper represents an ideal setting to answer this question. First, as will be shown later, with the exception of Hungary, trade credit accounts for at least roughly one-fourth of the total debt of a representative firm and about one-third of the short-term debt. Second, besides trade credit, alternative sources of finance are mostly unavailable to firms of the European countries that I focus on: the development of the stock and bond markets is modest.⁷ Then, to the extent that credit to firms are more likely to be rationed by financial institutions, the impact of collateral damage (as a consequence of the LTV ratio cap) will be magnified, and the net impact will be determined by the extent to which trade credit use offsets financial credit. According to the results, collateral damage caused debt use on secured basis to decrease in firms with high collateral value by 1.2 percentage points more than in firms with low collateral value, whereas it caused trade credit use to increase in firms with high collateral value by 0.2 percentage points more than in firms with low collateral value. Thus, this result provides a new evidence on how the corporate sector adjusts to collateral shocks through unsecured lending: firms that experienced a collateral damage after the LTV ratio cap turned to trade credit, allowing the corporate sector to mitigate the effects of such shocks.

There is an extensive literature that analyzes the role of collateral in determining firms' borrowing capacity. However, due to data limitations, almost all papers in this literature preclude the analyses of SMEs. One of the contributions of this paper then is to incorporate firm heterogeneity in this analysis. Using a sample that consists of both large firms and SMEs, I estimate the effect of asset tangibility as proxy for collateral on firms' borrowing capacity separately for different size groups. According to the estimation results, the impact of asset tangibility on debt capacity is pronounced across very small firms (i.e., micro enterprises⁸).

⁶There is an extensive literature of both theoretical and empirical papers explaining the existence of trade credit. Some emphasize a transaction motive for trade credit, while others emphasize a financial motivation. There are also many papers that have analyzed whether trade credit and bank loans are substitutes or complements. See Giannetti, Burkart, and Ellingsen (2011), Love (2011), and Uchida, Udell, and Watanabe (2013) for comprehensive reviews of this literature.

⁷Figure 4 shows the composition of financial liabilities of corporate sector for some European countries I focus on. The percentages are based on official statistics I obtain from Eurostat. According to these percentages, two meaningful sources of external finance are financial loans and trade credit, whereas the proportion of bonds in firm finance is limited.

⁸Micro enterprises are firms with employees less than 10.

This finding suggests that the interpretation of the role of asset tangibility in determination of firms' borrowing capacity should be done with caution: asset tangibility does not inversely measure the extent of financing constraints, but rather measures firms' ability to pledge collateral. Firms' ability to pledge collateral increases firms' borrowing capacity to the extent that tangible assets on firms' balance sheet are liquid (e.g., Campello and Giambona (2013)).

There is also a growing literature that studies the effectiveness of macroprudential policies.⁹ In this literature, there are several papers that provide evidence on whether LTV and DTI¹⁰ ratio caps are effective in mitigating the negative effects of housing boom by controlling credit growth and asset price inflation (e.g., Crowe, DellAriccia, Igan, and Rabanal (2013), Kuttner and Shim (2013)). While suggestive, these studies come with many caveats. Due to data limitations and/or identification struggles, they are not able to clarify the channels through which an LTV cap reduces systemic risk. This paper highlights one micro-level mechanism through which LTV caps on mortgages limit the positive feedback between asset price inflation and firms' capacity to borrow. Doing so, this paper provides a new evidence on how macroprudential policies targeting one sector–such as LTV ratio caps in household sector–might result in an unintentional consequence in another sector–such as collateral damage in corporate sector. This is an important caveat that policy makers should consider when implementing macroprudential policy.

The paper proceeds as follows. Section 2 reviews the policy experiments with LTV ratios. Section 3 presents the details about data. Section 4 explains the methodology. Section 5 presents empirical analysis. Section 6 concludes.

2 Policy Experiments with LTV Ratio Cap

In principle, macroprudential policies aim to limit the risk of widespread disruptions to the provision of financial services and thereby minimize the impact of such disruptions on real economy as a whole (Lim (2011)). Among others, LTV, LTI and DSTI caps have increasingly been implemented to reduce systemic risk generated by strong credit growth and credit-driven asset price inflation during boom-bust episodes. The caps are viewed as having macro-prudential impact through restricting the quantity of credit by limiting the funding available for certain borrowers to dampen growth in asset prices. In addition, they enhance the resilience of both the banks and their borrowers. Figure 1 depicts the transmission channels of a tightening of the LTV, LTI and DSTI limits.

The caps on LTV ratios are particularly popular in Asian and European countries. According to a survey conducted by the IMF in 2010, 20 out of 49 countries use caps on LTV ratios as a macro-prudential instrument. Specifically, among 20 countries, 11 countries set fixed caps while 9 countries adopt time-varying caps (Lim (2011)).

LTV ratio cap is a cap on the ratio of the value of the loan (L) relative to the value of the underlying collateral (V). LTV ratio cap imposes a limit on borrowers' capacity to borrow on collateralized lending. LTV ratios are not harmonised under the Capital Requirements Directive (CRD)/Capital Requirements Regulation (CRR), and rather are implemented at national discretion. Therefore,

⁹See Claessens (2015), Claessens et al. (2013), and Lim (2011) for a comprehensive review of existing studies.

¹⁰Debt to Income.

LTV ratio cap can be viewed as a recommendation or restriction of credit standards that banks should follow when issuing loans. Nevertheless, the explicit LTV limits vary both across types of loan within a country as well as across countries. The LTV limit in an individual country is usually related to the type of loan (commercial versus residential) and currency of the loan (domestic versus foreign currency) with foreign currency mortgages usually being subject to stricter LTV limits. Furthermore, the heterogeneity among countries also stems from differences in the coverage of institutions to which the explicit LTV limit is applied.

I undertake a detailed investigation for this particular policy action. The policy action data set used in this paper draws on a variety of sources. I use data sources from several studies developed by Borio and Shim (2007), Claessens, Ghosh, and Mihet (2013), Claessens (2015), Crowe, DellAriccia, Igan, and Rabanal (2013), Hilbers, Otker-Robe, Pazarbasioglu, and Johnsen (2005), Kuttner and Shim (2013), Lim (2011), Lim, Krznar, Lipinsky, Otani, and Wu (2013), and Srobona, Jaromír Benes, Lund-Jensen, Schmieder, and Severo (2011). Wherever available, I also use the official documents from central banks and supervisory & regulatory authorities including their annual reports and financial stability reports, press releases, email responses from these institutions. I use these secondary sources to cross-check them with the information given in the papers. Doing so, I obtain full and accurate information on relevant policy actions. The final data set allows me to precisely identify not only the coverage, but also the implementation date of LTV ratio policy in the sample of European countries: Bulgaria, Hungary, The Netherlands, Norway, Sweden, and Turkey.

I use Table 1 as reference to generate the dummy variable of difference-in-difference estimations, which captures the introduction and/or the tightening of LTV ratio cap on mortgages in the given country.

Country	Authority	Dates Active
Bulgaria	Bulgarian National Bank	June 2004–December 2007
Hungary	Magyar Nemzeti Bank	March 2010–present
Netherlands	Rijksoverheid (Dutch government)	2007-present
Norway	Norges Bank	March 2010–present
Sweden	Finansinspektionen & Sveriges Riksbank	2010-present

Table 1: LTV Ratio Policy Experiments

NOTES: Table 1 reports both the time period and the national authority that associated with the LTV ratio cap policies of the countries I study.

Turkey

The Central Bank of Republic of Turkey December 2010-present

In the below, I briefly document the background that motivates LTV ratio policy action as well as the details regarding the implementation of this policy in the given country.

• Bulgaria: Credit to households grew rapidly during transition to EU accession. A credit boom was accompanied by a house price boom in early 2000s. Towards mid 2000s, while the credit risk in corporate sector stabilized, it continued to accelerate in consumers and mortgage segments. Thus, Bulgarian National Bank introduced LTV ratio caps on residential real estate mortgages. To be exact, in June 2004, it introduced of a 70% LTV ratio for mortgages risk-weighted at 50%; and in April 2006 the risk weighting for mortgage loans used in the

calculation of the capital adequacy ratio is effectively raised by lowering LTV ratio from 70 % to 50 %.

- Hungary: In 2010, a large share of mortgage loans was provided in foreign currency which made unhedged borrowers in the household sector vulnerable to exchange rate volatility. To address the excessive foreign exchange rate lending to households, the authorities took some LTV ratio policy actions. To illustrate, in March 2010, the maximum LTV ratio was set at 75, 60 and 45 % for forint, euro and other foreign currency loans, respectively. The relevant LTV limits are somewhat higher for vehicle financing loans and residential real estate leasing (80, 65 and 50 % respectively for forint, euro and other foreign currency loans). This limit applies to all institutions providing financial services in Hungary.
- The Netherlands: The recent Dutch housing-market slump follows a long period of very rapid growth in property prices. Between 1985 and 2007, house prices rose by a cumulative 228%, while consumer prices increased only 56%. Dutch demand for houses was also boosted by government policy. Traditionally, the Dutch government has pursued a policy of promoting home ownership. This limited access to social housing and continued rise in house prices encouraged households including low-income earners. Banks were quite willing to lend to this group, even at very high LTV ratios (the average LTV ratio was 114 % in 2007). Additionally, mortgage interest rate were very attractive. They declined from 7.12 % in 1995 to 4.82% in 2007. Market sentiment changed over the course of 2007, triggered by the sub-prime mortgage problems in the U.S. Dutch banks also started to question lending standards. Before 2007, banks had substantial leeway in their lending to households. However, in 2007, the banks signed up to "Mortgage Lenders' Code of Conduct." In 2011, for all financial intermediaries under this code-of-conduct supervision, the code was tightened introducing a LTV ratio of 104% (plus transfer tax) on mortgages. In combination, LTV limit for new mortgage loans decreases stepwise 1 percentage point per annum from 106% in 2012 to 100% in 2018.
- Norway: Household debt (mainly mortgages) reached a high level and was a key risk in Norway. Low unemployment and wealth effects from increases in oil prices helped to boost the accumulation of household debt. Lax lending standards and aggressive mortgage lending practices also played a key role. To address the problem of housing debt, in March 2010, Norges Bank set LTV limit at 90%. According to law, LTV ratios on home equity loans should generally not exceed 75 %. Further, in December 2011, the authority tightened the law by lowering LTV ratio on mortgages to 85%, and lowering LTV ratio on home equity loans to 70 %.
- Sweden: The Swedish mortgage market is large. Since the mid-1990s, housing prices in Sweden have risen and Swedish households' mortgage debts have increased substantially. In 2001, mortgages comprised 30 % of the Swedish banks total lending secured on housing. Even if Swedish housing prices and indebtedness of Swedish households temporarily dampened after the global financial crisis of 2008–2009, they have subsequently continued to rise. To address the risks of growing household indebtedness and rising housing prices, Sveriges Riksbank

introduced a mortgage cap in 2010. According to general guideline of Financial Stability authority (Finansinspektionen), LTV ratio is at 85 % of a property's value. It applies to all credit institutions providing mortgages, but only covers new loans.

• **Turkey**: After the global financial crisis, Turkey observed a rapid increase in domestic demand and credit growth, and increased foreign currency borrowing by banks. In late 2010, the Central Bank of Republic of Turkey applied limits on mortgages in order to curb credit growth and increase credit quality, LTV ratio cap of 75 % on housing loans to consumer, LTV ratio cap of 50% on purchases of commercial real estate.

Figure 2 shows that the introduction of LTV ratio cap on mortgages led to a slowdown in price inflation in housing markets of the countries I focus on. This policy shock also led to a slowdown in prices in commercial markets. For instance, Figure 3 shows before-after LTV ratio policy difference in commercial price index by regions of Sweden. According to this figure, price of commercial buildings decreased in all Swedish regions with the exceptions of Blekinge and Sdermanland after the introduction of LTV ratio cap in Swedish housing market. I observed such a correlation between the price movements of residential property and commercial property due to the fact that both commercial and residential property use and compete for the same fixed supply of land (DiPasquale and Wheaton (1996)). Therefore, the timing of LTV ratio cap in housing market presents an ideal source of identification to investigate how firms' debt financing decisions respond to price shocks to the value of firm collateral. I will discuss this in detail in section 4.

3 Data

3.1 Firm-level Data

In my analysis, I use cross-country firm-level data from ORBIS. ORBIS is a commercial data set compiled by Bureau Van Dijk (BvD) that provides administrative data on millions of firms in Europe. The financial and balance sheet information is initially collected by local Chambers of Commerce and in turn, is relayed to BvD through some 40 different information providers.

The data set has financial accounting information from detailed, harmonized balance-sheets, income statement and profit/loss accounts of financial and non-financial firms. This data set is crucially different from other data sets that are commonly-used in the literature such as COMPU-STAT for the United States, COMPUSTAT Global, and Worldscope databases, since 99 percent of the companies in ORBIS are private, whereas former data sets contain mainly information on large listed companies. In ORBIS, only less than 2 percent of the firms are publicly listed (which is also separately marketed under the product called OSIRIS). As stated in Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015), there are several inherent biases affecting the download process, and a number of irregularities in the raw data, which will result in large data loss unless they are dealt with. Then, I fully follow their detailed instructions in order to construct a database that is nationally representative with minimal missing information.

In order to show how representative the data I use in this paper is, I refer to Table 2.¹¹ This table shows how much of the official gross output data from Eurostat is covered by the firms in ORBIS–AMADEUS data for the total economy for a sample of European countries. These countries refer to the countries with policy experiments of LTV ratio cap on mortgages: Bulgaria, Hungary, The Netherlands, Norway, and Turkey. Each cell is the ratio of value of total output produced by the firms in ORBIS–AMADEUS data relative to value of total output produced as in the official data. For a given country-year, ratios are computed by taking the ratio of aggregated gross output values where aggregated gross output is computed by totalling gross output over common available sectors for which the gross-output related variable is available in both data sets.¹² Missing ratios still appear in some country-year due to missing Eurostat data. As shown in Table 2 with the exception of Netherlands, ORBIS–AMADEUS data can account more than 50 percent of the aggregate output in all countries.

The sample I use in this paper is mainly composed of micro (1–9 employees), small (10–49 employees) and medium (50–249 employees) enterprises that account for a significant fraction of economic activity in Europe and the majority of economic activity in the sample of selected European countries mentioned above. In Table 3, each cell corresponds to the share of indicated size category's number of firms in total economy from the relevant data source for the given country in 2010. Number of firms is summed over overlapping sectors with Eurostat SBS data. This table illustrates that the sample is broadly representative in terms of size distribution. This feature is an important difference of this paper relative to the literature that works with both financial and real variables at the firm level. Most of this literature focuses on listed firms that account for less than 1 percent of the observations in the sample.

The main financial variables used in the analysis are total assets, sales, tangible fixed assets, components of debt, cash holdings, inventory, and earnings before interest, taxes, depreciation, and amortization (EBITDA). I transform financial variables to real using CPI with 2005 base and converting to dollars using the end-of-year 2005 dollar/national currency exchange rate. The data set has detailed sector classification (up to four-digit NACE Rev. 2 industry classification). I drop financial firms, real-estate and government-owned firms, and use all the other sectors.¹³

I use two different samples in analysis: Full Sample and Permanent Sample. Full sample contains all firms that are present in the database for at least three years before LTV ratio cap policy, and one year when LTV ratio cap is in place. This sample includes unbalanced panels from the following countries with the relevant periods given in parentheses: Bulgaria (1997–2013), Hungary (1997–2012), Netherlands (1997–2012), Norway (2004–2013), Sweden (1998–2013) and Turkey (2003–2012). Permanent sample covers firms from the Full sample without non-consecutive yearly observations (i.e., which appear, disappear and reappear in the sample). This sample includes balanced panels from the following countries with the relevant periods given in parentheses: Hungary (2004–2012),

¹¹This table is reproduced from Table 6.1 and Table 6.2 of Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015).

¹²See Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015) for further details on the construction of percentages.

¹³I also drop firms operating in the sectors outside SNA production boundary (NACE Rev. 2 sectors T & U).

Netherlands (2000–2012), Norway (2004–2013), Sweden (1999–2012) and Turkey (2005–2012).¹⁴

Tables 5–6 (C.1–C.2) show the percentages of firms by firm type and country in the Full (Permanent) sample. The firms in each country's sample refer to ones with non-missing value of the variable on which the percentages are based. In the table, each cell corresponds to the share of indicated category's number of firms in total economy of the given country-period (%). In the first two panels, shares are constructed based on firm size, and firm size is measured by the logarithm of real total assets and the number of employees, respectively. In the bottom panel, firms are categorized based on firm age.

3.2 Variable Definitions

The measures of debt and firm controls that I examine in capital structure regressions are coming from the intersection of influential papers on the topic over the last two decades. In this section, I firstly provide a detailed discussion about the measures of debt, secondly I define the variables I use as firm controls in the empirical analysis.

There are differences in the composition of total debt, so the type of firm debt analyzed should base on the objective of the analysis. The objective of this analysis is to investigate how changes in firm collateral are related to changes in firm capital structure decisions. The components of debt have different relationship with firm collateral, for example trade credit use is higher in firms with low levels of pledgable assets whereas secured debt use is higher in firms with high levels of pledgable assets. The most appropriate way to analyze different debt measures would be firstly to separate total debt as "secured debt" and "unsecured debt." I do not have matched data at firm-bank level, but I can still make a plausible distinction between secured and unsecured debt. For example, "Loans" and "Trade Creditors," which are sub-accounts of current liabilities in the balance sheet can be treated as secured debt and unsecured debt, respectively. Because loans that are provided by financial institutions heavily require collateral, whereas trade credit that is provided by suppliers does not require collateral.

In addition, total debt includes items like income tax payable, social expenditure payable, pension fund provisions, which are used for other purposes rather than financing, so it may overstate the amount of financial debt. Such items are recorded in balance sheet under the account called "Other Liabilities." However, this account also covers other items such as "Other Short-term Debt," "Other Short-term Creditors," and "Other Long-term Non-Interest Bearing Debt," which are all used for financing purposes. Thus, excluding such types of debt from total debt might underestimate the amount of financial debt. None of these items are reported under "Other Liabilities" as in separate sub-accounts. Given this caveat in reporting of sub-accounts, therefore, in order to avoid any estimation errors, I use different alternative measures as follows: TotDebt: The sum of shortterm and long-term debt; FinDebtTOL: Total debt excluding trade credit; FinDebt: Total debt excluding trade credit and other liabilities; STFinDebt: Short-term debt from financial institution; STFinDebtSTOL: Short-term Debt excluding trade credit, and TC: Trade Credit.¹⁵

¹⁴I apply different cleaning steps and quality checks before constructing these two samples. The details regarding all this procedure is available in the appendix.

¹⁵The details on the composition of liabilities are available in the appendix.

The determinants of firm financing decisions I use in the empirical analysis are the ones commonly used in the related literature. To proxy *Collateral*, I consider all types of pledgable assets that firms are able to post as collateral in loan/credit applications. In balance sheets, pledgable assets refer to total book value of tangible fixed assets ("PP&E") which are composite of net book value of land and building, net book value of machinery and equipment, and book value of other tangible assets such as plant and equipment in progress and leased assets. Thus, I define asset tangibility as the ratio of total book value of tangible fixed assets to total book value of assets and use this to measure collateral in leverage analysis. *Profitability* is the ratio of EBITDA to book value of total assets. This variable is used to measure internal finance. To proxy growth opportunities, I use Sales Growth, which is defined as the logarithmic difference of real sales (measured in 2005 constant dollars).¹⁶ Size is logarithm of book value of total assets (measured in 2005 constant dollars). Age is the logarithm of (1 + firm age) where firm age in period t is defined as t minus the date of incorporation plus one. To proxy firm liquidity, I use *Cash*, which corresponds to the ratio of cash and cash equivalents to book value of total assets. Inventory corresponds to total inventories (raw materials+in progress+finished goods), and is normalized by book value of total assets. Table 7 shows the descriptive statistics on all these variables used in the empirical analysis. In general there is a good deal of variation that allows me to show both economic and econometric inferences I study in this paper.

4 Identification

In this paper, I aim to investigate how firms' debt financing decisions respond to changes in collateral value. I focus on a sample of European countries with policy experiments of LTV ratio caps on mortgages (Bulgaria: 2004–2007, Hungary: 2010-present, The Netherlands: 2008-present, Norway: 2010-present, Sweden: 2010-present, and Turkey: 2010-present). Their policy experience provides an ideal setting for identification. Tightening of LTV ratios on mortgages affects house prices, a policy shock that is plausibly exogenous to any individual firm. This policy shock has a general equilibrium effect through the demand and supply of houses on collateral values, and it is affecting firm financing decisions through collateral channel.

My hypothesis is that if LTV ratio cap is effective in curbing borrowers' demand by tightening borrowers' capacity to borrow, this will have a negative effect on real estate prices, then firms who had higher valued collateral pre-LTV policy shock should experience a bigger drop after-LTV policy shock in their secured borrowing relative to firms who had lower valued collateral pre-LTV policy shock.

¹⁶Most studies in the related literature use Tobin's Q (measured as the ratio of market value of total assets to book value of total assets) to proxy profitable growth opportunities. Their analysis bases on large US firms who report their cash flow statements reported to COMPUSTAT, so those firms have information on market values. Here, I study capital structure of private firms of different sizes, and private firms do not have information on market values of assets/equity. Thus, Sales Growth is the most appropriate measure for private firms to proxy profitable growth opportunities. Similar to Sales Growth, the ratio of intangible assets to total assets can be considered as an alternative proxy for Tobin's Q. Intangible assets include R&D and advertising expenses and firms are more likely to increase such expenses when they have profitable growth opportunities. Given the limited number of firms reporting information on intangible assets in the data, I instead use Sales Growth in my empirical analysis.

In order to test this hypothesis, I develop a difference-in-difference estimation including countryyear, sector-year, firm fixed effects. This estimation, by interacting firm collateral with a time dummy that separates the period before and after the LTV ratio policy shock, captures before-after shock difference in secured borrowing of firms with similar collateral on their balance sheet prior LTV policy shock. The inclusion of fixed effects is important. These absorb the impact on firms' debt financing decisions of changing country and sector conditions and factors driving both aggregate and local demand. In particular, firm fixed effects will control for unobserved, time-invariant firm characteristics. For example, if high risk taker firms hold less collateral on average, such firms might be affected differentially from LTV policy shock. In the estimation, this average affect will be fully absorbed by firm fixed effects, and will not invalidate the identification. Further, industry fixed effects at four-digit-level sector codes will absorb time-varying demand conditions, because most of demand fluctuations derive from country- and industry-specific factors, not from firm-specific factors. Any remaining variation in firm specific demand conditions does not vary systematically by the collateral value. The benchmark differences-in-differences equation is:

$$y_{i,s,c,t} = \beta_1 \text{Collateral}_{i,s,c} \times \text{LTV}_{c,t} + \beta_2 X_{i,s,c,t} + \mu_{c,t} + \mu_{s,t} + \mu_i + \varepsilon_{i,s,c,t},$$
(4.1)

where the indices i,s,c,t denote a firm, a sector, a country and a year, respectively. I use different debt measures as dependent variables: TotDebt, FinDebtTOL, and TC. X is a matrix containing standard control variables: Sales Growth, Profitability, Size, Inventory, and Cash (see section 3.2 for further details on the construction of variables).

Collateral is a dummy variable that equals one if asset tangibility is higher than the median of the distribution of this variable. I prefer using this variable as in the form of the dummy variable for two reasons. First, given the interaction specification, indicator variable makes the interpretation of the coefficient straightforward by identifying the group of interest clearly. Second, to avoid concerns about selection into becoming high collateral holder a consequence of the policy, Collateral is a predetermined firm-level dummy which bases on the value of asset tangibility three years prior to the application of LTV ratio cap.¹⁷ LTV_{c,t} a dummy variable that equals to 1 in the year(s) when LTV ratio cap is in place in country c (see reference years given in Table 1). The interaction variable is the simple multiplication of these two dummy variables.

I include $\mu_{s,t}$ that captures country-year fixed effects, $\mu_{s,t}$ that controls sector-year fixed effects where sectors are classified according to four digit NACE Revision 2 codes. μ_i capture firm-specific effects, and $\varepsilon_{i,s,c,t}$ is the error term. By using firm fixed effects I will be identifying solely from firm changes over time. Therefore, I can not identify the main effect of Collateral which is absorbed by firm fixed effects because Collateral is a predetermined firm-level dummy variable. The level (direct) effect of policy shock to LTV ratio is absorbed by country-year fixed effects as other time

¹⁷Changes in collateral level from low to high in any years during LTV ratio cap is in place is 13 percent of observations. Collateral is a dummy variable that equals one if asset tangibility is higher than 0.15 at any time during three years prior to the application of LTV ratio cap. 0.15 corresponds to the median of the distribution of asset tangibility.

fixed effects. Both sector-year and country-year fixed effects will absorb the effects of any other industry and country level shocks as well as the effects of any year.

In this specification, the coefficient of interest is β_1 . It captures the treatment effect of LTV ratio cap and equals the DD estimate. It multiplies the interaction term, which can be interpreted as dummy variable equal to one for the firms who are exposed to the treatment when LTV ratio cap is in place. Given the fact that the LTV ratio policy affects the firms with differential collateral values differently, the coefficient β_1 allows one to measure before-after shock difference in the corresponding debt measure in firms with high collateral value relative to before-after shock difference in firms with collateral value.

5 Results

5.1 Reconciling Results with Firm Capital Structure Literature

In this section, I first would like to verify that my results are consistent with those of existing papers studying the determinants of firm capital structure decisions. Using a comprehensive panel data that consists of both large firms and SMEs, I revisit stylized capital structure regressions. This exercise represents an important attempt because existing evidence bases mostly on the sample of large publicly traded firms operating in developed countries. Then, I estimate equation (4.1) without LTV interaction:

$$y_{i,s,c,t} = \beta_1 X_{i,s,c,t} + \mu_{c,t} + \mu_{s,t} + \mu_i + \varepsilon_{i,s,c,t},$$
(5.1)

where the indices i,s,c,t denote a firm, a sector, a country and a year, respectively. I use different debt measures as dependent variables: TotDebt, FinDebt, FinDebtTOL, and TC. X is a matrix containing standard control variables: Collateral, Sales Growth, Profitability, Size, Inventory, Cash, and Age (see section 3.2 for further details on the construction of variables).

The above equation (4.1) includes fixed effects. Specifically, firm fixed effects will control for unobserved, time-invariant heterogeneity across firms. Industry-year fixed effects at four-digit-level sector codes will absorb time-varying sector specific conditions. Country-year fixed effects will absorb time-varying country specific conditions. As mentioned earlier, since most of the aggregate demand fluctuations derive from country- and industry-specific factors, not from firm-specific factors, country-, industry-year fixed effects will also absorb fluctuations in aggregate demand that might drive the relationship between firm debt financing decisions and any firm controls.

Table 10 reports the estimation results. In order to see whether unobserved heterogeneity drives the results or not, I firstly estimate equation 5.1 without firm- and industry fixed effects. I use standard leverage measure i.e., TotDebt in the regressions. Columns (1)-(3) show that the estimators pass "fixed-effects stress tests" of Lemmon, Roberts, and Zender (2008) because all firm-level controls are still statistically significant after the inclusion of fixed effects.¹⁸ This result verifies

¹⁸Lemmon, Roberts, and Zender (2008) argue that the traditional firm-level controls in capital structure decisions become largely insignificant in explaining the variation in firm leverage when the model accounts for time invariant

that traditional determinants in capital structure decisions have ability in explaining the variation of leverage both in cross section and within the firm in the time series.

As noted in section 3.2, there are differences in the composition of total debt, so the type of firm debt analyzed should base on the objective of the analysis. The objective of this analysis is to investigate how changes in firm collateral are related to changes in firm capital structure decisions. The components of debt have different relationship with firm collateral, for example trade credit use is higher in firms with low levels of pledgable assets whereas debt use from financial institutions is higher in firms with high levels of pledgable assets. Therefore, I estimate equation 5.1 individually for different debt measures.

Column (3)–(6) correspond to debt measures i.e., total debt (TotDebt), total debt excluding trade credit (FinDebtTOL), total debt excluding trade credit and total other liabilities (FinDebt), and trade credit (TC), respectively. As dependent variables in the estimation, they are all normalized by total assets. Further details on the composition of debt measures are given in the appendix. ¹⁹

The results mirror previous work on related literature. The positive and statistically significant coefficient on "Collateral" in Columns (3)–(5) suggest that if a large fraction of a firm's assets are tangible, those assets can be pledged as collateral diminishing the risk of agency costs on debt. Therefore, firms can issue more debt given the lenders be more willing to supply funds. On the other hand, negative coefficient on collateral in Column (6) suggests that trade credit use is lower in firms with higher levels of collateral.²⁰ Trade credit is an expensive form of finance, so firms with higher levels of collateral appear to use more from other sources of finance.

According to the results, more profitable firms have lower debt of any form, consistent with theoretical predictions in the literature. According to pecking order theory, firms prefer internal funds rather than debt since internal funds have no adverse selection problem (Myers and Majluf (1984)). In other words, highly profitable firms use less debt (more internal equity).

Further, I find a positive and statistically significant coefficient on Sales Growth, suggesting that growing firms use higher debt of all types in order to take advantage of investment opportunities they face. This finding follows Kalemli-Ozcan, Laeven, and Moreno (2015). Using a comprehensive sample of European firms,²¹ they show that in the run-up to the crisis, a typical European firm increasingly issues debt to utilize profitable investment opportunities (proxied by Sales Growth). The relevant literature is not able to find such a positive relationship between investment opportunities and debt financing. The existing papers mostly utilize Compustat data, representing only a typical

firm effects.

¹⁹As it can be inferred from Table 7, there is a significant number of zero observations in terms of bank loans (both short-term and long-term financial loans i.e., STFinDebt, LTFinDebt). In case FinDebt is used as dependent variable in the estimation, the dependent variable is censored from left, and thus tobit model would be rather an appropriate one. However, within a tobit model, I can not control for μ_i and $\mu_{s,t}$ by means of a dummy variable approach (incidental parameters problem), and no tobit model analogous to the "fixed-effects" logit exists. Honoré (1992) has proposed a "fixed effects" tobit that does not impose distributional assumptions. However, it is hard to implement, and partial effects can not be estimated. I therefore do not try his approach. Alternatively, I estimate tobit model of benchmark leverage regression only with country dummies and compare it with simple pooled OLS. The inferences from these two models are similar.

 $^{^{20}}$ Trade credit is negatively correlated with collateral levels, supporting the implications of the theoretical model developed in Cunat (2007).

²¹The structure of their sample is similar to the one I use in this paper since their sample is constructed based on ORBIS–AMADEUS, as in this paper.

large publicly traded company in US, thus fail to provide evidence on SME finance.

Firm size has been empirically found to be positively related to capital structure. Most of the studies in the literature use cross-sectional variation and interpret the positive coefficient on "Size" as larger firms are highly leveraged. However, in this paper, I use within-firm variation and interpret the same coefficient as firms get bigger, they increase debt. The results show a positive coefficient for all types of debt except TotDebt and FinDebtTOL (Column (3)–(4)). As noted in section 3.2, these two measures are the most comprehensive debt measures due to the inclusion of total other liabilities (TOL). This account includes items like income tax payable, social expenditure payable, pension fund provisions, which are used for other purposes rather than financing, thus the extent of such items in these two measures might affect the relationship.

I also study the usage of trade credit within firms by including additional control variables. The result in column (6) shows that firms use more trade credit when they have higher level of inventories, reflecting a positive correlation between firm activity and trade credit use. The result on the "Inventory" variable also could be related to the use of inventories as collateral. The negative coefficient on "Cash" variable suggests that firms increase trade credit use when they face additional liquidity needs.²²

The first two columns provide results where age is also an explanatory variable in benchmark capital structure equation. In all other regressions, age is not available because it is a firm specific linear time trend, and is absorbed by firm and year fixed effects. Given the caveat in interpreting the coefficient on age in the regressions without firm fixed effects, the negative coefficient on age suggests that as firms age, they issue more equity, but less debt.

The economic effects of firm-level determinants of capital structure decisions are reported in square brackets under standard errors in columns (3)–(6) of Table 10. The relevant percentages highlight the economic importance of firm controls as determinants of firm debt, indeed collateral (proxied by asset tangibility) appears to be the key determinant of debt of any form. For example, in column (4), the economic effect of collateral is displayed in terms of percentage change in debt to its sample mean as each regressor increases from the 25th to the 75th percentile (1-IQR change), while all other variables are kept at their sample mean. To be precise, a 1-IQR change in firm collateral leads debt (measured by FinDebtTOL) to increase by 0.089, which is a 17.43% increase relative to the sample mean debt of 0.51.

5.2 Collateral and Firm Leverage: The Impact of LTV Ratio Cap Policy

As discussed in section 4 in detail, my hypothesis is that if LTV ratio cap is effective in curbing borrowers' demand by tightening borrowers' capacity to borrow, this will have a negative effect on real estate prices, then firms who had higher valued collateral pre-LTV policy shock should experience a bigger drop after-LTV policy shock in their secured borrowing relative to firms who had lower valued collateral pre-LTV policy shock. In order to test this hypothesis, I estimate

 $^{^{22}}$ In unreported results, I rerun column (6) by including the measures of short-term financial debt (e.g., STFinDebt and STFinDebtSTOL). The negative coefficients on those measures show that firms use more trade credit when they have lower level of short-term finance in other forms i.e. bank loans, reflecting that trade credit can serve as a substitute for short-term financial debt.

equation (4.1).

Table 11 shows the main results. According to the results in column (1), I find a significant collateral damage effect on firm leverage (TotDebt) after LTV ratio cap. This effect is different for firms with differential collateral values: LTV ratio cap decreased leverage in firms with high collateral value by 0.9 percentage point more than in firms with low collateral value.

In order to fully assess the impact of collateral damage on firms' financing decisions, all sources of external finance must be considered. One type of lending might substitute for another type of lending, one type of lender might substitute for another type of lender. Trade credit usage is immanent. As shown in Table 4, with the exception of Hungary, trade credit accounts for at least roughly one-fourth of the total debt of a representative firm and about one-third of the short-term debt. Second, besides trade credit, alternative sources of finance are mostly unavailable to firms of the European countries that I focus on: the development of the stock and bond markets is modest. Then, did firms that experienced a collateral damage after LTV ratio cap turn to trade credit as an alternative source of finance?

To the extent that credit to firms are more likely to be rationed by financial institutions, the impact of collateral damage (as a consequence of LTV ratio cap) will be magnified, and the net impact will be determined by the extent to which trade credit use offsets financial credit. According to the results in columns (2)-(3), collateral damage caused debt use on secured basis to decrease in firms with high collateral value by 1.2 percentage points more than in firms with low collateral value, whereas it caused trade credit use to increase in firms with high collateral value by 0.2 percentage points more than in firms with low collateral value.

Columns (5)-(6) correspond to the debt measures divided by total debt as the dependent variable. The evolution of these variables show the relative changes with respect to other debt sources. The results suggest that LTV ratio cap decreased the proportion of secured debt use (FinDebtTOL) in total debt in firms with high collateral values by 0.5 percentage point more than in firms with low collateral values, whereas it increased the proportion of trade credit use (TC) in total debt in firms with high collateral values by 0.5 percentage point more than in firms with low collateral values. In combination, these results verify the predictions I just stated above.

Does the amount of cash on hand influence trade credit use in the years when LTV ratio cap is in place? If a firm views trade credit as an alternative but expensive source of finance, I should find cash-rich firms increase trade credit to a smaller extent. Therefore, I estimate equation (4.1) using cash as an indicator of liquidity to test this hypothesis. Cash is a predetermined firm-level dummy which bases on the ratio of cash holding to total assets three years prior to the application of LTV ratio cap.²³ The pre-LTV level of cash is absorbed by firm fixed effects, and thus I can only observe the differential responses to LTV ratio cap. The results reported in columns (4) and (7) suggest that cash-rich firms increase their reliance on credit from suppliers to a smaller extent in the years when LTV cap is in place.

As shown in Table C.6 in the appendix, results are not driven by entry and exit into the sample,

 $^{^{23}}$ Cash is a firm-level dummy variable that equals one if the ratio of cash holdings to total assets is higher than 0.34 at any time during three years prior to the application of LTV ratio cap. 0.34 corresponds to the 75th of the distribution of this variable.

and are robust to consider a continuous sample of firms (see section 3.1 for details on the construction of permanent sample). In Table 12, I also conduct a placebo test using years prior to LTV ratio cap as the policy years (Bulgaria: 2002–2003, Hungary: 2005–2006, The Netherlands: 2005–2006, Norway: 2008–2009, Sweden: 2005–2006, and Turkey: 2008–2009) and I can not find that firms change their debt financing decisions as they do after-LTV policy shock.

5.3 Collateral and Firm Leverage: Average Effects by Different Size Deciles

So far I work with a linear specification to identify the impact of asset tangibility on firm debt financing. This specification delivers useful insights for the average firm in the sample. However, there are issues I need to account for while studying firm capital structure. As mentioned earlier, the industrial structure of the economies studied in this paper are tilted toward SMEs (see e.g., Table 3) and SME finance is more complex than large firm finance (Berger and Udell (1998), Berger and Udell (2006)).²⁴ For this reason, I turn to a specification where the effect of firm collateral on debt is estimated nonlinearly. This is done with a regression of the form:

$$y_{i,s,c,t} = \beta_1 \text{Collateral}_{i,s,c,t} + \sum_{k=2}^{10} \beta_k \times D_{k,t} \times \text{Collateral}_{i,s,c,t} + \sum_{k=11}^{20} \beta_k \times D_{k,t} + \beta_{11} X_{i,s,c,t} + \mu_{c,t} + \mu_{s,t} + \mu_{k,t} + \mu_i + \varepsilon_{i,s,c,t}, \qquad (5.2)$$

where the indices i,s,c,t denote a firm, a sector, a country and a year, respectively. X is a matrix containing standard control variables: Sales Growth, Profitability, Size. $D_{k,t}$ is a time-varying dummy variable that takes value 1 for all firms that fall in decile k of the size distribution in the given year t. In this way, collateral effect is estimated separately for each size class. Further, $D_{k,t}$ should be interpreted as size-year fixed effects will control for all the time varying differences between firms of different size.

Table 13 corresponds to dependent variables i.e. FinDebt, FinDebtTOL, and TC. Collateral is the ratio of tangible fixed assets to total assets. Collateral×size decile=k is the additional effect of collateral over and above the baseline effect for first decile firms captured by the variable collateral, and are reported in the first columns. Latter columns report the overall effect of collateral for a firm of decile k. The corresponding p-value from an F test with the null hypothesis that this effect is zero is given in the square parentheses in bold.

For secured debt obligations (FinDebt, FinDebtTOL), the results show that the impact of asset tangibility on debt capacity is pronounced across all firms of different sizes, but less pronounced across very small and large firms. This finding follows the conventional wisdom. Large firms are typically old, reputable, and less vulnerable to imperfections in credit markets, and hence they borrow with higher LTV ratios (lower collateral) in private debt markets since lenders generally consider

²⁴For example, the finance of very small firms with no track record and no collateral rely on insider funds (from start-up team, family, friends), trade credit and/or angel finance. As firms grow and accumulate collateral and track record, they access to intermediated finance from both equity and debt markets (i.e venture capital and loans from financial institutions). Large firms of known risk and track record issue commercial paper and/or obtain funds from public equity and debt markets in addition to loans from financial institutions.

them as low-risk borrowers.²⁵ For example, Berger and Udell (1998, 1995) with US data show that loans to low-risk borrowers are less likely to be collateralized. Similarly, Jimenez, Salas, and Saurina (2006) with Spanish data provide direct evidence of negative association between collateral and a borrower's risk.

The results for very small firms²⁶ suggest that the interpretation of the role of asset tangibility in determination of firms' borrowing capacity should be done with caution: asset tangibility does not inversely measure the extent of financing constraints, but rather measures firms' ability to pledge collateral. Firms' ability to pledge collateral increases firms' borrowing capacity to the extent that tangible assets on firms' balance sheet are liquid (e.g., Campello and Giambona (2013)).

Finally, columns (5) and (6) correspond to unsecured debt obligations (TC). The results show that the impact of asset tangibility on debt capacity is pronounced across all firms of different sizes. However, I do not observe a strong cross-sectional variation as I do in case of secured borrowing. The results suggest that firm heterogeneity (captured by firm size) does not play a strong role in determining trade credit use in firms with similar collateral on their balance sheet. This finding can be explained by the possibility that trade creditors act as "relationship lenders." Trade creditors have proprietary information about their customers and they are better positioned to repossess and resell the supplied goods (e.g., Mian and Smith (1992) McMillan and Woodruff (1999))), thus trade creditors might have an advantage over other lenders in providing credit to firms of all sizes including SMEs.

5.4 Collateral and Firm Leverage: Heterogenous Responses to LTV Ratio Cap Policy

In light of the results I discussed in the previous section, my hypothesis is that if the impact of asset tangibility on debt capacity is less pronounced across very small and large firms, I argue that firms of medium size deciles ("SME") should experience a bigger drop in their secured borrowing relative to firms of bottom and top size deciles ("VerySmall–Large") who had similar collateral damage on their balance sheet after-LTV policy shock.

In order to test this hypothesis, I turn to a triple differences-in-differences specification. I justify this specification by the use of medium-year fixed effects that capture all time varying differences between "SME" firms and "VerySmall–Large" firms. The estimation:

$$y_{i,s,c,t} = \beta_1 \text{SME}_{i,s,c} \times \text{Collateral}_{i,s,c} \times \text{LTV}_{c,t} + \beta_2 \text{Collateral}_{i,s,c} \times \text{LTV}_{c,t} + \beta_2 X_{i,s,c,t} + \mu_{\text{SME},t} + \mu_{c,t} + \mu_{s,t} + \mu_i + \varepsilon_{i,s,c,t},$$
(5.3)

²⁵In the sample, on average, large firms are 28,5 years old whereas the others in the lower deciles are 14,6 years old. The difference between these means is significant at the 1 percent level.

 $^{^{26}}$ The firms in bottom deciles (up to 7th decile) correspond to the majority of micro enterprises (0–9 employees), whereas the firms in the middle deciles (7th–9th deciles) correspond to the majority of SMEs (10–249 employees) in the sample. Furthermore, the firms in the bottom deciles are on average younger (13,6 years old) than the firms in the middle deciles (18,5 years old). The difference between these means is significant at the 1 percent level.

where the indices i,s,c,t denote a firm, a sector, a country and a year, respectively. X is a matrix containing standard control variables (Sales Growth, Profitability, Size for TotDebt and FinDebt-TOL; Sales Growth, Profitability, Size, Inventory and Cash for TC). To avoid selection concerns, I also use predetermined firm-level dummy to define "SME" firms.²⁷ $\mu_{\text{sme},t}$ are sme-year fixed effects.

Triple interaction term in equation (5.3) turns out to be important for identification. To see why, I compare the interpretation of coefficients in equation (4.1) to those of equation (5.3). In equation (4.1), β_1 captures the treatment effect of LTV ratio cap for typical firm holding high level of collateral. This is not the case for β_1 in equation (5.3) because now this coefficient reflects the treatment effect only for a typical "SME" firm with similar collateral value. Therefore, to understand how the treatment effect of LTV ratio cap varies with firm size, one should the compare β_1 to β_2 . For example, β_1 compared to β_2 would be the incremental effect of being "SME" firms during the year(s) in which LTV ratio cap is in place.

Table 14 reports the estimation results. The results verify the validity of the hypothesis stated above. For example, the results in column (1) show that LTV ratio cap decreased leverage in "SME" firms with high collateral value by 1.2 percentage points more than in "VerySmall–Large" firms with high collateral value. However, firms of different size did not behave differently in terms of trade credit use after they experienced a collateral damage (e.g., in column (3), β_2 is very small i.e., -0.0002). Table 15 shows that the results are not driven by "Large" ²⁸ firms, and are robust to the exclusion of large firms.

5.5 Aggregate Implications

The results presented in the previous sections suggest a significant collateral damage effect on firms' debt financing decisions after LTV policy shock. In this section, I will conduct a back-of-envelope calculation to link micro estimates I obtained from the difference-in-difference estimation to the actual corporate leverage patterns observed in the aggregate data. Doing so, if there observed a decline in aggregate corporate leverage after LTV policy shock, I will thus be able to gauge how the collateral damage effect contributed to this decline.

To construct aggregate measures, I use official statistics from Eurostat. Eurostat provides country-level balance sheets that have information on non-financial assets, financial assets and financial liabilities. However, the accounts from Eurostat's balance sheets are structured differently than those I have from firm-level balance sheets. In order to precisely compare the firm-level measures from ORBIS–AMADEUS data with the aggregate measures from Eurostat data, I work on a detailed correspondence of the accounts from these two data sets.²⁹

I cannot proceed with the analysis of aggregate implications of LTV policy using the pooled

 $^{^{27}}$ SME equals one if the given firm's size (measured by logarithm of real total assets) is between $75^{th}-95^{th}$ percentiles of the distribution at any time during the three years prior to the introduction of LTV ratio cap.

²⁸ "Large" firms refer to firms of top size deciles

²⁹Statistics on financial balance sheets come from Eurostat. To construct "Total Financial Liabilities" for nonfinancial corporations, I sum of F3: "Securities other than shares," F4: "Loans," "F6:Insurance premiums", and F7: "Other accounts receivable/payable." This summation would correspond to "TotDebt" in ORBIS-AMADEUS. Next, to construct "Total Assets," I sum of F_AS: "Financial Assets," and T11: "Total Fixed Assets, (net)." This summation would correspond to "TOAS" in ORBIS-AMADEUS. See http://ec.europa.eu/eurostat/data/database?node_ code=nasa_f_bs for further details

sample of six European countries since Eurostat does not provide full information for some countries in the pooled sample.³⁰ Then, I proceed with the analysis of Sweden, which has the better coverage in both data sets. I rewrite equation (4.1) using benchmark leverage measure i.e. TotDebt (the ratio of total debt to total assets):

$$TotDebt_{i,s,t} = \beta_1 Collateral_{i,s} \times LTV_t + \beta_2 X_{i,s,t} + \mu_i + \mu_{s,t} + \varepsilon_{i,s,t},$$
(5.4)

where the indices i,s,t denote a firm, a sector and a year, respectively. X is a matrix containing standard control variables: Sales Growth, Profitability, Size.

In order to sum of collateral damage effect across all Swedish firms (without grouping them based on their collateral values), I first use Collateral as continuous firm-level variable, which is defined by the ratio of tangible fixed assets to total assets. Second, to avoid concerns that the share of tangible fixed assets in total assets might have been changed as a consequence of the policy, I compute firm-level average of this ratio for the 1998–2007 period that excludes all three years prior the introduction of LTV ratio cap in 2010. In the estimation, I cannot identify the main effect of Collateral because Collateral is proxied by the firm-level average ratio of tangible assets to total assets, which is absorbed by firm fixed effects. The level (direct) effect of policy shock to LTV ratio is absorbed by sector-year fixed effects as other time fixed effects. Sector-year fixed effects will absorb the effects of any other industry shocks as well as the effects of any year.

Third, I sort all firms in ascending order based on Collateral. I denote the before-after LTV policy shock difference in firm leverage as Δ TotDebt_{*i*,*s*}. Then, based on equation (5.4), the difference in the before-after LTV policy difference in firm leverage in two consecutive firms is expressed in the below:

$$\Delta \text{TotDebt}_{i,s} - \Delta \text{TotDebt}_{i-1,s} = \beta_1(\text{Collateral}_{i,s} - \text{Collateral}_{i-1,s}), \quad (5.5)$$

Further, I define the aggregate effect of LTV policy shock as:

$$\Delta \text{TotDebt} = \sum_{i \ge 0} \omega_{i,s} \Delta \text{TotDebt}_{i,s}, \qquad (5.6)$$

where $\omega_{i,s}$ indicates the share of tangible fixed assets of firm *i* in aggregate tangible fixed assets.³¹ The empirical methodology I use estimates the differential effect of LTV policy shock across firms

³⁰For instance, Eurostat does not provide information on non-financial assets for Bulgaria, Norway and Turkey, which prevents me from computing aggregate total assets.

³¹Statistics on the components total non-financial assets come from Eurostat. They are disaggregated by industry based on NACE Revision 2. Before, summing the values over sectors, I exclude non-overlapping sectors that are not used in the analysis: K: "Financial and insurance activities," L: "Real estate activities," T: "Activities of households as employers; undifferentiated goods- and services-producing activities of households for own use," and "U: Activities of extraterritorial organisations and bodies." To construct "Tangible Fixed Assets (Net)," I take the difference between T11: "Total Fixed Assets (Net)" and T112: "Intangible Fixed Assets." See http://ec.europa.eu/eurostat/data/database?node_code=nama_t20_21_c for further details.

with different collateral values. In order to pin down the level effect of this policy shock, I assume that the before-after LTV policy shock difference in firm leverage equals zero for the firm with the lowest collateral value. Then, I have this equality:

$$\Delta \text{TotDebt}_{i,s} = \beta_1(\text{Collateral}_{i,s} - \text{Collateral}_0) \text{ for } i > 0, \qquad (5.7)$$

Thus, I find the lower bound of aggregate effect of LTV policy shock on firm leverage by estimating the below equation:

$$\Delta \text{TotDebt}_{i,s} = \beta_1 \sum_{i \ge 0} \omega_{i,s} (\text{Collateral}_{i,s} - \text{Collateral}_0).$$
(5.8)

According to the estimation results of equation (5.4), β_1 equals to -0.032 (see Table 16). Further, based on calculations, second term in RHS of equation (5.8) equals to 0.026. Then, aggregate effect of the LTV ratio cap in Swedish corporate sector is -0.08% (=-0.032*0.026).

Based on aggregate statistics from Eurostat, I compute the average before-after LTV policy shock difference in aggregate corporate leverage (i.e., the ratio of aggregate corporate financial liabilities to aggregate corporate assets³²) over the 1998–2013 period. It equals to -0.5% (=0.351-0.356). Thus, the LTV ratio cap that resulted in collateral damage explains 16% (=0.08%/0.5%) of the decline in aggregate corporate leverage.

6 Conclusion

This paper investigates how firms' capital structure decisions respond to changes in the value of firm collateral, caused by real estate price shock. Through the collateral channel, shocks to the value of real estate can have a significant impact on the firms' borrowing capacity. I provide evidence on this mechanism by using LTV ratio caps on mortgages in a number of European countries as policy shocks that affect real estate prices. In the analysis, I conduct a difference-in-difference exercise using a unique and comprehensive micro panel data covering both large firms and SMEs. This allows me to better identify and quantify the effects of policy shocks to the value of firm collateral on debt financing by distinguishing them from local demand shocks and local general equilibrium effects. I find a significant collateral damage effect on firms' debt financing decisions: LTV ratio cap caused secured debt to decrease in firms with high collateral value more than in firms with low collateral value.

Further, I investigate how shocks to the value of collateral affect firms' choice between secured and unsecured debt financing. I find that firms that experienced a collateral damage after LTV ratio cap turned to trade credit as an alternative source of finance: collateral damage caused trade credit use to increase in firms with high collateral value more than in firms with low collateral value. These findings document a new evidence on how firms adjust to collateral shocks through trade

 $^{^{32}\}mathrm{See}$ footnote 29 for details on the construction of this ratio

credit use.

I believe that this paper has important implications on the role of collateral pledging in transmitting boom-bust cycles in real estate markets to the corporate sector. The inclusion of SMEs in the analysis is crucial given the structure of European economies. Europe consists of bank-dominant economies tilted toward externally dependent SMEs. Among all sources of external financing, European firms typically prefer debt financing to fund working capital and/or investment. European banks heavily prefer real-estate as collateral especially for SMEs. In order to minimize dependency of SME finance on collateral pledging, and broaden SMEs' access to funding, policy makers should develop alternative policies. For example, SME loan guarantee schemes enable SMEs to borrow more than would otherwise be possible based on their collateral. They might help mitigating the aggregate effects of collateral damage caused by real estate price shock.

This paper also highlights that macroprudential policies in one sector–such as LTV ratio caps on mortgages targeting household sector–might result in an unintentional consequence in another sector–such as collateral damage in corporate sector. This is an important caveat that policy makers should consider when implementing macroprudential policy.

References

- ACHARYA, V. V., T. EISERT, C. EUFINGER, AND C. W. HIRSCH (2014): "Real Effects of the Sovereign Debt Crisis in Europe: Evidence from Syndicated Loans," *CEPR Discussion Paper No. DP10108.*
- BERGER, A. N., AND G. F. UDELL (1995): "Relationship Lending and Lines of Credit in Small Firm Finance," *Journal of Business*, pp. 351–381.

(1998): "The Economics of Small Business Finance: The Roles of Private Equity and Debt Markets in the Financial Growth Cycle," *Journal of Banking & Finance*, 22(6), 613–673.

- (2006): "A More Complete Conceptual Framework for SME Finance," Journal of Banking & Finance, 30(11), 2945–2966.
- BERNANKE, B. S., AND M. GERTLER (1995): "Inside the Black Box: The Credit Channel of Monetary Policy Transmission," *NBER Working Paper, No:5146.*
- BORIO, C. E., AND I. SHIM (2007): "What Can (Macro-) Prudential Policy do to Support Monetary Policy?," *BIS Working paper, No:42.*
- CAMPELLO, M., AND E. GIAMBONA (2013): "Real Assets and Capital Structure," Journal of Financial and Quantitative Analysis, 48(05), 1333–1370.
- CHANEY, T., D. SRAER, AND D. THESMAR (2012): "The Collateral Channel: How Real Estate Shocks Affect Corporate Investment," *The American Economic Review*, 102(6), 2381–2409.
- CHOI, W. G., AND Y. KIM (2005): "Trade Credit and the Effect of Macro-Financial Shocks: Evidence from US Panel Data," *Journal of Financial and Quantitative Analysis*, 40(04), 897–925.
- CLAESSENS, S. (2015): "An Overview of Macroprudential Policy Tools," Annual Review of Financial Economics, 7(1).
- CLAESSENS, S., S. R. GHOSH, AND R. MIHET (2013): "Macro-prudential Policies to Mitigate Financial System Vulnerabilities," *Journal of International Money and Finance*, 39, 153–185.
- CROWE, C., G. DELLARICCIA, D. IGAN, AND P. RABANAL (2013): "How to Deal with Real Estate Booms: Lessons from Country Experiences," *Journal of Financial Stability*, 9(3), 300–319.
- CUNAT, V. (2007): "Trade Credit: Suppliers as Debt Collectors and Insurance Providers," *Review* of Financial Studies, 20(2), 491–527.
- CVIJANOVIĆ, D. (2014): "Real Estate Prices and Firm Capital Structure," *Review of Financial Studies*, pp. 35–83.
- DIPASQUALE, D., AND W. C. WHEATON (1996): Urban Economics and Real Estate Markets, vol. 23. Prentice Hall Englewood Cliffs, NJ.

- GAN, J. (2007): "Collateral, Debt Capacity, and Corporate Investment: Evidence from a Natural Experiment," *Journal of Financial Economics*, 85(3), 709–734.
- GIANNETTI, M., M. BURKART, AND T. ELLINGSEN (2011): "What You Sell IS What You Lend? Explaining Trade Credit Contracts," *Review of Financial Studies*, 24(4), 1261–1298.
- HILBERS, P., I. OTKER-ROBE, C. PAZARBASIOGLU, AND G. JOHNSEN (2005): "Assessing and Managing Rapid Credit Growth and the Role of Supervisory and Prudential Policies," *IMF Working Paper*.
- HONORÉ, B. E. (1992): "Trimmed LAD and Least Squares Estimation of Truncated and Censored Regression Models with Fixed Effects," *Econometrica*, pp. 533–565.
- JIMENEZ, G., V. SALAS, AND J. SAURINA (2006): "Determinants of Collateral," *Journal of Finan*cial Economics, 81(2), 255–281.
- KALEMLI-OZCAN, S., L. LAEVEN, AND D. MORENO (2015): "Debt Overhang, Rollover Risk and Investment in Europe," *Working Paper*.
- KALEMLI-OZCAN, S., B. SORENSEN, C. VILLEGAS-SANCHEZ, V. VOLOSOVYCH, AND S. YESILTAS (2015): "How to Construct Nationally Representative Firm Level Data from the ORBIS Global Database," *NBER Working Paper*, No:21558.
- KIYOTAKI, N., AND J. MOORE (1997): "Credit Cycles," *The Journal of Political Economy*, 105(2), 211–248.
- KUTTNER, K. N., AND I. SHIM (2013): "Can Non-interest Rate Policies Stabilize Housing Markets? Evidence from a Panel of 57 Economies," *NBER Working Paper, No:19723.*
- LEMMON, M. L., M. R. ROBERTS, AND J. F. ZENDER (2008): "Back to the Beginning: Persistence and the Cross-Section of Corporate Capital Structure," *The Journal of Finance*, 63(4), 1575–1608.
- LIM, C. H. (2011): "Macroprudential Policy: What Instruments and How to Use Them? Lessons from Country Experiences," *IMF Working Paper*.
- LIM, C. H., I. KRZNAR, F. LIPINSKY, A. OTANI, AND X. WU (2013): "The Macroprudential Framework: Policy Responsiveness and Institutional Arrangements," *IMF Working Paper*.
- LOVE, I. (2011): "Trade Credit versus Bank Credit During Financial Crises," Trade Finance During the Great Trade Collapse, The World Bank, Washington, DC, pp. 27–39.
- MATEUT, S., S. BOUGHEAS, AND P. MIZEN (2006): "Trade Credit, Bank Lending and Monetary Policy Transmission," *European Economic Review*, 50(3), 603–629.
- MCMILLAN, J., AND C. WOODRUFF (1999): "Interfirm Relationships and Informal Credit in Vietnam," *Quarterly Journal of Economics*, pp. 1285–1320.

- MIAN, S. L., AND C. W. SMITH (1992): "Accounts Receivable Management Policy: Theory and Evidence," *The Journal of Finance*, 47(1), 169–200.
- MYERS, S. C., AND N. S. MAJLUF (1984): "Corporate Financing and Investment Decisions When Firms Have Information that Investors Do not Have," *Journal of Financial Economics*, 13(2), 187–221.
- NANDA, R., AND T. NICHOLAS (2014): "Did Bank Distress Stifle Innovation During the Great Depression?," *Journal of Financial Economics*, 114(2), 273–292.
- NILSEN, J. H. (2002): "Trade Credit and the Bank Lending Channel," Journal of Money, Credit and Banking, 34(1), pp. 226–253.
- SROBONA, M., S. I. JAROMÍR BENES, K. LUND-JENSEN, C. SCHMIEDER, AND T. SEVERO (2011): "Toward Operationalizing Macroprudential Policies: When to Act?," *IMF Global Financial Stability Report.*
- UCHIDA, H., G. F. UDELL, AND W. WATANABE (2013): "Are Trade Creditors Relationship Lenders?," Japan and the World Economy, 25, 24–38.

7 Tables & Figures

YEAR	BG	HU	NL	NO	SE
1999		0.79	0.17	0.63	0.52
2000		0.84	0.21	0.63	0.56
2001		0.71	0.2	0.77	0.6
2002	0.57	0.73	0.22	0.79	0.63
2003	0.6	0.66	0.21	0.65	0.65
2004	0.79	0.76	0.22	0.67	0.67
2005	0.83	0.8	0.23	0.69	0.68
2006	0.84	0.81	0.25	0.67	0.71
2007	0.91	0.79	0.25	0.71	0.7
2008	0.94	0.75	0.28	0.59	0.73
2009	0.92	0.87	0.31	0.78	0.87
2010	0.96	0.76	0.35	0.79	0.88
2011	0.88	0.72	0.3	0.72	0.82
2012	0.88	0.82	0.28	0.67	0.76

Table 2: COVERAGE IN TOTAL ECONOMY BASED ON GROSS OUTPUT

NOTES: Table 2 presents the ratios that are calculated based on gross output. The total sample consists of firms that report data with positive values of the corresponding measure (i.e. gross-output). The country codes within these classifications are as follows: BG (Bulgaria), HU (Hungary), NL (Netherlands), NO (Norway), HU (Hungary) and SE (Sweden). BvD provides firm-level information on gross-output for all sectors of a given European country between 1999–2012, however Eurostat SBS data provides information on gross output with the exceptions of some sectors. So, for a given country-year, total economy percentages are computed by taking the ratio of aggregated gross output values where aggregated gross output is computed by totalling gross output over these sectors for which gross-output related variable is available in both data sets. For further details on the construction of percentages, see Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015).

	DO			NO	an	
	BG	ΗU	NL	NO	SE	TR
PANEL A: EUROSTAT SBS						
Micro	91.2	94.6	93.6	91.6	94.5	n.a.
SMEs	8.6	5.2	6.2	8.2	5.4	n.a.
LARGE	0.2	0.1	0.2	0.2	0.2	n.a.
	BG	HU	NL	NO	SE	TR
PANEL B: ORBIS-AMADEUS						
Micro	90.9	77.7	84.8	91.2	91.2	47.9
SMEs	8.9	21.6	14.7	8.2	8.2	46.4
LARGE	0.2	0.7	0.5	0.6	0.6	5.7

Table 3: SIZE DISTRIBUTION IN TERMS OF TOTAL ECONOMY, 2010

NOTES: In Table 3, each cell corresponds to the share of indicated size category's number of firms in total economy from the relevant data source for the given country in 2010 (%). Number of firms is summed over overlapping sectors with Eurostat SBS data. In each panel, the first three rows report the percentages from ORBIS-AMADEUS and the next three rows are the same percentages from Eurostat's SBS data. Each column is a different country with the following codes: BG (Bulgaria), NL (Netherlands), NO (Norway), HU (Hungary), RO (Romania), SE (Sweden), and TR (Turkey). Micro corresponds to firms with employees less than 10, SMEs corresponds Small and Medium Enterprises with employees between 10 and 249, and Large corresponds to firms with 250 employees or more. For further details on the construction of percentages, see Kalemli-Ozcan, Sorensen, Villegas-Sanchez, Volosovych, and Yesiltas (2015).

Table 4: Relative size of Trade Credit	(AVERAGE OVER FIRMS)

Country	Dataset	Trade credit/assets (%)	Trade credit/total debt (%)	Trade credit/Short term debt (%)
Bulgaria	ORBIS-AMADEUS	19	33	43
Hungary	ORBIS-AMADEUS	3	4	6
Netherlands	ORBIS-AMADEUS	12	19	27
Norway	ORBIS-AMADEUS	15	20	26
Sweden	ORBIS-AMADEUS	11	19	26
Turkey	ORBIS-AMADEUS	23	40	47
United States	NSSBF	17	35	50

NOTES: ORBIS–AMADEUS, Bureau Van Dijk database contains data on firms of all sizes for the given European country; NSSBF, National Survey on Small Business Finance 4630 small and medium US firms. ORBIS–AMADEUS covers years 1997-2013 for Bulgaria and Hungary; 1997-2012 for Netherlands; 2004-2012 for Norway; 1998-2013 for Sweden; 2003-2012 for Turkey, and NSSBF is a 1998 cross-section

Country	Pooled	BG	HU	NL	NO	SE	TR
Period	1997-2013	1997-2013	1997-2013	1997-2012	2004-2012	1998-2013	2003-2012
PANEL A: TOTAL ASSETS							
All	285,204	7,924	54,772	5,212	75,302	137,829	4,165
Small	75.0	83.8	81.2	3.5	72.5	77.8	18.4
	(213, 858)	(6, 643)	(44, 492)	(181)	(54, 570)	(107, 207)	(765)
Medium	20.1	13.5	15.5	26.4	23.6	19.4	44.3
	(57, 338)	(1,069)	(8,508)	(1, 376)	(17,740)	(26,798)	(1, 847)
LARGE	4.9	2.7	3.2	70.1	4.0	2.8	37.3
	(14,008)	(212)	(1,772)	(3,655)	(2,992)	(3,824)	(1,553)
Panel B: Employment							
All	235,792	3,850	28,033	4,958	60,456	136,847	1,648
Micro	73.0	47.4	57.9	11.2	70.1	81.0	14.5
	(172,086)	(1, 824)	(16, 242)	(554)	(42, 408)	(110, 819)	(239)
SMEs	25.5	42.1	39.9	68.5	29.1	18.4	67.9
	(60,089)	(1,620)	(11, 174)	(3,397)	(17, 622)	(25, 157)	(1,119)
LARGE	1.5	10.5	2.2	20.3	0.7	0.6	17.6
	(3, 617)	(406)	(617)	(1,007)	(426)	(871)	(290)

Table 5: PERCENTAGE OF FIRMS IN FULL SAMPLE-BY FIRM TYPE AND COUNTRY

To be Continued.

Country	Pooled	BG	HU	NL	NO	SE	TR
Period	1997-2013	1997-2013	1997-2013	1997-2012	2004-2012	1998-2013	2003-2012
Panel C: Age							
All	279,728	3,152	54,142	5,203	75,293	137,773	4,165
Infant	10.2 (28,568)	14.1 (443)	9.8 $(5,297)$	2.9 (150)	17.2 (12,979)	6.9 (9,481)	5.2 (218)
Adolescent	11.4 (31,833)	21.5 (677)	13.5 (7,310)	4.5 (235)	12.5 (9,409)	10.1 (13,915)	6.9 (287)
Middle-aged	69.6 (194,745)	49.0 (1,543)	76.7 (41,519)	43.7 (2,272)	70.2 (52,870)	68.0 (93,731)	67.5 (2,810)
Old	8.8 (24,582)	15.5 (489)	0.0 (16)	48.9 (2,546)	0.0 (35)	15.0 (20,646)	20.4 (850)

Table 6: PERCENTAGE OF FIRMS IN FULL SAMPLE-BY FIRM TYPE AND COUNTRY

NOTES: Tables 5–6 show the percentages of firms by firm type and country. The firms in each sample refer to ones with non-missing value of the variable on which the percentages are based. In both tables, each cell corresponds to the share of indicated category's number of firms in total economy of the given country-period (%). In Table 5, shares are constructed based on firm size, and firm size is measured by the logarithm of real total assets and the number of employees, respectively. In Table 6, firms are categorized based on firm age. In Table 5, firm size categories are constructed based on predetermined dummies that each equals one if the firm satisfies the criterion for the corresponding firm category at any time during the three years prior to the introduction of LTV ratio cap: SMALL equals one if the given firm's size is below 75^{th} percentile of the distribution, MEDIUM equals one if the given firm's size is above 95^{th} percentile of the distribution. MICRO equals one if the given firm has employees less than 10, SMEs equals one if the given firm has employees between 10 and 249, and LARGE equals one if the given firm has employees higher than 250. In Table 6, INFANT equals one if the given firm's age is between 3–4, MIDDLE-AGED equals one if the given firm's age is between 5–24, and OLD equals one if the given firm's age is 25 or above. Numbers in parentheses refer to the total number of firms with non-missing value of the variable on which the percentages are based.

Sample: Full Period: 1997–2013 Countries: BG, HU, NL, NO, SE, TR					
VARIABLE	Mean	Median	STD. DEV.	25тн Рст.	75тн Рст.
TotDebt	0.62	0.61	0.36	0.4	0.8
FinDebt	0.13	0	0.2	0	0.22
FinDebtTOL	0.51	0.48	0.34	0.29	0.67
STFinDebt	0.022	0	0.058	0	0
STFinDebtSTOL	0.32	0.27	0.24	0.15	0.44
TC	0.11	0.057	0.13	0.0067	0.16
Collateral	0.26	0.15	0.26	0.035	0.43
Sales Growth	0.015	0.0079	0.47	-0.12	0.15
Profitability	0.12	0.12	0.19	0.038	0.22
Size	13	13	1.6	12	14
Cash	0.21	0.13	0.22	0.031	0.33
Inventory	0.16	0.05	0.22	0	0.25
Age	2.6	2.7	0.64	2.2	3

Table 7: Descriptive Statistics: Full Sample

NOTES: Table 7 reports descriptive statistics of main variables used in the empirical analysis for Full Sample. Debt measures are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit; FinDebt: Total debt excluding trade credit and other liabilities; STFinDebt: Short-term debt from financial institution; STFinDebtSTOL: Short-term Debt excluding trade credit, and TC: Trade Credit. Debt measures are all divided by total assets. Further details on the composition of debt measures are given in the appendix. Collateral is the ratio of total tangible fixed assets to total assets. Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Age is the logarithm of (1+firm age) where firm age in period t that is defined as t minus the date of incorporation plus one. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets.

Table 8: Composition of Liabilities-By Firm Type

SAMPLE: Full								
Period: 1997–2013								
COUNTRIES: BG, HU, NL, NO, SE, TR								
	A	All	SM	IALL	Me	DIUM	LA	RGE
	Mean	<u>Median</u>	Mean	<u>Median</u>	Mean	<u>Median</u>	Mean	<u>Median</u>
(% of Total Liabilities)								
STFinDebt	0.03	0	0.023	0	0.045	0	0.063	0
TC	0.11	0.051	0.1	0.04	0.12	0.08	0.11	0.075
STOL	0.54	0.53	0.57	0.58	0.47	0.43	0.46	0.42
LTFinDebt	0.14	0	0.13	0	0.18	0	0.12	0
LTOL	0.11	0	0.11	0	0.095	0	0.16	0.051

NOTES: Table 8 reports descriptive statistics of debt measures by different firm size groups. Debt measures are defined as follows. STFinDebt: Short-term debt from financial institutions; TC: Trade Credit, STOL: Other Short-term Liabilities, LTFinDebt: Long Term Interest Bearing Debt, LTOL: Other Long-term Liabilities, and TC: Trade Credit. Further details on the composition of debt measures are given in the appendix.

Table 9: Cash and Collateral Holdings-By Firm Type

Sample: Full								
Period: 1997–2013								
COUNTRIES: BG, HU, NL, NO, SE, TR								
	A	All	SM	IALL	Me	DIUM	LA	ARGE
	Mean	Median	Mean	<u>Median</u>	Mean	<u>Median</u>	Mean	Median
(% of Total Assets)								
Collateral	0.26	0.15	0.25	0.13	0.28	0.18	0.27	0.2
Inventory	0.16	0.05	0.15	0.03	0.17	0.09	0.17	0.12
Cash	0.22	0.13	0.25	0.17	0.15	0.083	0.092	0.035

NOTES: Table 9 reports descriptive statistics of cash and collateral holdings by different firm size groups. Collateral is the ratio of total tangible fixed assets to total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Debt measures are all divided by total assets.

Dep. Var.:	TotDebt (1)	TotDebt (2)	TotDebt (3)	FinDebtTOL (4)	FinDebt (5)	$\begin{array}{c} \mathrm{TC} \\ (6) \end{array}$
Collateral	0.156^{***} (0.002)	0.160^{***} (0.002)	0.157*** (0.003) [6.58%] [10.00%]	0.225*** (0.000) [11.47%] [17.43%]	0.281*** (0.002) [56.20%] [85.38%]	-0.080*** (0.001) [-18.91%] [-28.15%]
Profitability	-0.567*** (0.004)	-0.536*** (0.004)	-0.352*** (0.003) [-10.79%] [-10.33%]	-0.274*** (0.005) [-10.21%] [-9.78%]	-0.092*** (0.001) [-13.45%] [-12.88%]	-0.068*** (0.001) [-11.13 %] [-11.31%]
Sales Growth	$\begin{array}{c} 0.074^{***} \\ (0.001) \end{array}$	0.071^{***} (0.01)	0.043*** (0.001) [3.26%] [1.87%]	0.026*** (0.001) [2.40%] [1.38%]	0.003*** (0.000) [1.08%] [0.62%]	0.017*** (0.000) [7.11 %] [4.17%]
Size	-0.019*** (0.001)	-0.021*** (0.001)	-0.009*** (0.001) [-2.32%] [-2.90%]	-0.013*** (0.001) [-4.08%] [-5.10%]	0.032*** (0.002) [39.38%] [49.23%]	0.003*** (0.000) [4.36 %] [5.45%]
Inventory						0.034*** (0.002) [6.49%] [7.73%]
Cash						-0.042*** (0.001) [-8.40 %]
Age	-0.050^{***} (0.002)	-0.049^{***} (0.001)				[1100,0]
Observations R^2	$1,555,196 \\ 0.17$	$1,555,196 \\ 0.21$	$1,581,186 \\ 0.82$	$1,581,186 \\ 0.77$	$1,581,186 \\ 0.81$	$1,513,978 \\ 0.77$
Firm Fixed-Effects	no	no	yes	yes	yes	yes
Sector Fixed-Effects Country Fixed-Effects Year Fixed-Effects Sector×year Fixed-Effects	no yes yes no	yes yes no	no no no yes	no no no yes	no no no yes	no no no yes
Country×vear Fixed-Effects	no	no	ves	ves	ves	ves

SAMPLE: Full Period: 1997-2013 COUNTRIES: BG, HU, NL, NO, SE, TR

NOTES: Table 10 reports the results of the estimation of equation (5.1). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, FinDebt, and TC. They are defined as follows. TotDebt: The sum of shortterm and long-term debt; FinDebtTOL: Total debt excluding trade credit; FinDebt: Total debt excluding trade credit and other liabilities, and TC: Trade Credit. Debt measures are all divided by total assets. Further details on the composition of debt measures are given in the appendix. Collateral is the ratio of total tangible fixed assets to total assets. Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Debt measures are all divided by total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. The first figures in square brackets under the t-statistics represent percentage changes in leverage relative to the sample as each continuous regressor increases by its standard deviation, while all other regressors are kept at their sample mean. In the same manner, the second figures in **bold** represent percentage changes in leverage relative to the sample mean as each continuous regressor increases from 25^{th} to the 75^{th} percentiles, while all other regressors are kept at their sample mean ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

Sample: Full Period: 1997–2013

COUNTRIES: BG, HU, NL, NO, SE, TR

Dependent Variable:	TotDebt	FinDebtTOL (divided by to	TC otal assets)	TC	FinDebtTOL (divide	TC d by total d	TC ebt)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Collateral×LTV	-0.009*** (0.002)	-0.012^{***} (0.001)	0.002^{***} (0.000)	0.002*** (0.000)	-0.005^{***} (0.001)	0.005^{***} (0.001)	0.004*** (0.001)
Cash×LTV				-0.001^{**}			(0.071^{***})
				(0.000)			(0.002)
Firm Controls	yes	yes	yes	yes	yes	yes	yes
Number of observations \mathbf{R}^2	$1,581,186 \\ 0.81$	$\substack{1,581,186\\0.8}$	1,513,978 0.77	$1,513,978 \\ 0.77$	$1,581,186 \\ 0.74$	$1,513,978 \\ 0.73$	1,513,978 0.73
Firm Fixed-Effects	yes	yes	yes	yes	yes	yes	yes
Sector×year Fixed-Effects	yes	yes	yes	yes	yes	yes	yes
Country×year Fixed-Effects	yes	yes	yes	yes	yes	yes	yes

NOTES: Table 11 reports the results of the estimation of equation (4.1). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–4 and Columns 5–7, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is a dummy variable that equals one if asset tangibility is higher than its median at any time during three years prior to the application of LTV ratio cap. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total inventories (raw materials+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

PLACEBO TEST

Sample: Full

Period: 1997–2013 Countries: BG, HU, NL, NO, SE, TR

Dependent Variable:	TotDebt	FinDebtTOL (divided by to	TC otal assets)	TC	FinDebtTOL (divide	TC d by total d	TC ebt)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Collateral×LTV	0.003^{**} (0.001)	0.002^{**} (0.002)	0.0004 (0.001)	0.0002 (0.001)	0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
$Cash \times LTV$				-0.001^{**} (0.000)			0.001 (0.001)
Firm Controls	yes	yes	yes	yes	yes	yes	yes
Number of observations \mathbf{R}^2	$1,133,583 \\ 0.79$	$1,133,583 \\ 0.75$	$1,102,027 \\ 0.76$	$1,102,027 \\ 0.76$	$1,133,583 \\ 0.73$	1,102,027 0.72	$1,102,027 \\ 0.72$
Firm Fixed-Effects Sector×year Fixed-Effects	yes	yes	yes	yes	yes	yes	yes ves
Country×year Fixed-Effects	yes	yes	yes	yes	yes	yes	yes

NOTES: Table 12 reports the results of the estimation of equation (4.1). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–4 and Columns 5–7, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is a dummy variable that equals one if asset tangibility is higher than its median at any time during three years prior to the application of LTV ratio cap. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total inventories (raw materials+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

Countries: BG, HU, NL, NO, SE, TR							
Dependent Variable:	FinDebt		FinDel	otTOL	TC		
	(1) (2)		(3)	(4)	(5)	(6)	
Collateral	0.115***	0.115***	0.094***	0.094***	-0.036***	-0.036***	
	(0.004)	[0.00]	(0.007)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 2$	0.070***	0.185***	-0.002	0.092***	-0.017***	-0.053***	
	(0.004)	[0.00]	(0.01)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 3$	0.113***	0.228***	0.016**	0.110***	-0.031***	-0.067***	
	(0.004)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 4$	0.152***	0.267***	0.040***	0.134***	-0.042***	-0.078***	
	(0.005)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 5$	0.191***	0.304***	0.072***	0.166***	-0.051***	-0.087***	
	(0.005)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 6$	0.223***	0.336***	0.098***	0.192***	-0.055***	-0.091***	
	(0.005)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 7$	0.242***	0.355***	0.112***	0.206^{***}	-0.059***	-0.095***	
	(0.005)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 8$	0.246***	0.359***	0.110***	0.204^{***}	-0.061***	-0.097***	
	(0.005)	[0.00]	(0.01)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 9$	0.237***	0.350***	0.093***	0.187***	-0.057***	-0.093***	
	(0.005)	[0.00]	(0.008)	[0.00]	(0.003)	[0.00]	
$Collateral \times size decile = 10$	0.147***	0.138***	0.006	0.100***	-0.035***	-0.071***	
	(0.008)	[0.00]	(0.01)	[0.00]	(0.004)	[0.00]	
Firm Controls	yes		yes		yes		
Number of observations	1,581,186		1,581,186		1,581,186		
\mathbb{R}^2	0.77		0.72		0.77		
Firm Fixed-Effects	yes		yes		yes		
Sector×year Fixed-Effects	yes		yes		yes		
Country×year Fixed-Effects	yes		yes		yes		

SAMPLE: Full Period: 1997–2013 Countries: BG, HU, NL, NO, SE, TR

NOTES: Table 13 reports the results of the estimation of equation (5.2). The dependent variables are different debt measures i.e. FinDebtTOL, FinDebt, and TC. They are defined as follows. FinDebtTOL: Total debt excluding trade credit; FinDebt: Total debt excluding trade credit and other liabilities, and TC: Trade Credit. Debt measures are all divided by total assets. Further details on the composition of debt measures are given in the appendix. Collateral is the ratio of tangible fixed assets to total assets. Collateral×size decile=k is the additional effect of Collateral over and above the baseline effect for first decile firms captured by the variable Collateral, and are reported in the first columns. Latter columns report the overall effect of Collateral for a firm of decile k. The corresponding p-value from an F test with the null hypothesis that this effect is zero is given in the square parentheses in **bold**. The regression is run with decile specific time-varying time trends (an interaction of the size decile dummy and time trend). Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Debt measures are all divided by total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm. t-statistics are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedasticconsistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses of the first column. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

SAMPLE: Full Period: 1997–2013 Countries: BG, HU, NL, NO, SE, TR

Dependent Variable:	TotDebt	FinDebtTOL	TC	FinDebtTOL	TC	
	(div	ided by total as	sets)	(divided by total debt)		
	(1)	(2)	(3)	(4)	(5)	
SME×Collateral×LTV	-0.012^{***}	-0.012^{***}	-0.0002	-0.003**	0.003^{**}	
	(0.002)	(0.002)	(0.002)	(0.001)	(0.001)	
Collateral×LTV	-0.006***	-0.010***	0.003^{***}	-0.005***	0.004^{***}	
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	
Firm Controls	yes	yes	yes	yes	yes	
Number of Observations	1,581,186	1,581,186	1,513,978	1,581,186	1,513,978	
\mathbb{R}^2	0.81	0.8	0.77	0.74	0.73	
Firm Fixed-Effects	ves	Ves	Ves	ves	ves	
SME ver Fixed Effects	yes	yes	Ves	yes	yes	
Soutony year Fixed Effects	ycs	yes	yes	ycs	ycs	
C t t File File C	yes	yes	yes	yes	yes	
Country×year Fixed-Effects	yes	yes	yes	yes	yes	
<u>F-test</u>						
Collateral×LTV	0.00	0.00	0.00	0.00	0.00	

NOTES: Table 14 reports the results of the estimation of equation (5.3). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–3 and Columns 4–5, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is a dummy variable that equals one if asset tangibility is higher than its median at any time during three years prior to the application of LTV ratio cap. SME equals one if the given firm's size (measured by logarithm of real total assets) is between $75^{th}-95^{th}$ percentiles of the distribution at any time during the three years prior to the introduction of LTV ratio cap. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets, the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

SAMPLE: Full-Excluding Large Firms PERIOD: 1997–2013

COUNTRIES: BG, HU, NL, NO, SE, TR

Dependent Variable:	TotDebt	FinDebtTOL	TC	FinDebtTOL	TC	
	(div	ided by total as	(divided by total debt)			
	(1)	(2)	(3)	(4)	(5)	
CMEY Collectory by LTW	0.019***	0.019***	0.0007	0.002**	0.009**	
SME×Conaterai×L1V	-0.013	-0.012	-0.0007	-0.003	0.003	
	(0.002)	(0.002)	(0.01)	(0.001)	(0.001)	
Collateral×LTV	-0.006***	-0.010***	0.003^{***}	-0.004***	0.004^{***}	
	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	
Firm Controls	yes	yes	yes	yes	yes	
Number of Observations	1,496,146	1,496,146	1,429,489	1,496,146	1,429,489	
\mathbb{R}^2	0.82	0.8	0.77	0.74	0.74	
Firm Fixed-Effects	Ves	ves	ves	ves	ves	
SMExvear Fixed-Effects	ves	ves	ves	ves	ves	
Soutonyyear Fixed Effects	yes	yes	yes	yes	yes	
Sector × year Fixed-Effects	yes	yes	yes	yes	yes	
Country×year Fixed-Effects	yes	yes	yes	yes	yes	
<u>F-test</u>						
Collateral×LTV	0.00	0.00	0.00	0.00	0.00	

NOTES: Table 15 reports the results of the estimation of equation (5.3). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–3 and Columns 4–5, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is a dummy variable that equals one if asset tangibility is higher than its median at any time during three years prior to the application of LTV ratio cap. SME equals one if the given firm's size (measured by logarithm of real total assets) is between $75^{th}-95^{th}$ percentiles of the distribution at any time during the three years prior to the introduction of LTV ratio cap. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

SAMPLE:	Full
Period:	1998 - 2013

COUNTRIES: Sweden

Dependent Variable:	TotDebt (div	FinDebtTOL ided by total as	TC sets)	FinDebtTOL (divided by t	TC total debt)
	(1)	(2)	(3)	(4)	(5)
Collateral×LTV	-0.032*** (0.003)	-0.059^{***} (0.002)	0.024^{***} (0.001)	-0.037*** (0.002)	0.032^{***} (0.002)
Firm Controls	yes	yes	yes	yes	yes
Number of Observations \mathbf{R}^2	$1,041,289 \\ 0.80$	$1,041,289 \\ 0.76$	$1,030,895 \\ 0.74$	$1,041,289 \\ 0.70$	$1,030,895 \\ 0.70$
Firm Fixed-Effects Sector×year Fixed-Effects Country×year Fixed-Effects	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes
$\frac{\text{F-test}}{\text{Collateral} \times \text{LTV}}$	0.00	0.00	0.00	0.00	0.00

NOTES: Table 16 reports the results of the estimation of equation (5.4). The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–3 and Columns 4–5, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is constructed as firm-level average of the ratio of tangible fixed assets to total assets for the 1998–2007 period. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

Figure 1: TRANSMISSION CHANNELS OF A TIGHTENING OF THE LTV, LTI AND DSTI LIMITS





Figure 2: The Impact of LTV Ratio Cap on House Prices



(b) Source: OECD



(c) Source: OECD





Figure 3: BEFORE-AFTER LTV RATIO POLICY DIFFERENCE IN COMMERCIAL PRICE INDEX BY REGION IN SWEDEN (SOURCE: STATISTICS SWEDEN)



Figure 4: Composition of Firm Liabilities: Aggregated Data from Eurostat



- 15

10

2

0

Eurozone

Source: Eurostat, National Accounts Data

(b) Long Term Liabilities–2006



(c) Short Term Liabilities–2011

Netherlands

Short Term Loans

Hungary

Short Term Bonds

Trade Credits

(d) Long Term Liabilities-2011

Sweden

Appendix A Data Cleaning Process

The dataset I use in this paper combines firm-level information across different BvD products (ORBIS disk 2005, ORBIS disk 2009, ORBIS disk 2013, AMADEUS online 2010 fromWRDS, and AMADEUS disk 2014). I clean the data in four steps. First, I clean the raw data off basic reporting mistakes. Second, I restrict this data to the one I use in my analysis and verify the internal consistency of balance sheet information. Third, I apply further quality checks (paper specific cleaning steps) and construct two different samples: Full Sample and Permanent Sample. Lastly, in each of these samples, we winsorize the variables used in the analysis.

A.1 Steps to Clean Basic Reporting Mistakes

I implement the following steps to clean the raw data off basic reporting mistakes.

- 1. I drop firms if any detail of assets/liabilities/capital, and sales, operating revenue, wage bill and depreciation is negative in any year.
- 2. I drop firms if they report sales, operating revenue, total assets, the sum of shareholderfunds and liabilities, and shareholder's capital as zero in any year.
- 3. I drop firms if employment is either zero or if employment is negative or greater than 2 millions in any year.
- 4. I drop firms if any of total assets, total liabilities and shareholder funds is missing in all years they report data to BvD.

A.2 Steps to Construct the Main Sample

- For a given firm ORBIS-AMADEUS provides financial statements regarding different consolidation codes i.e. C1, C2, U1 and U2.³³ Given this fact, I dropped C2 accounts to avoid double accounting in our analysis.³⁴
- 2. I first drop firms with missing information regarding their industry of activity. Second, I drop financial, real-estate firms. I also drop firms operating in the sectors outside SNA production boundary (NACE Rev. 2 sectors T & U).
- 3. I drop state-owned firms.
- 4. I drop firms with missing information on the variables used in benchmark analysis.
- 5. I drop years with observations less than 1000 in a given country.

 $^{^{33}}$ C1: account of a company- headquarter of a group, aggregating all companies belonging to the group (affiliates, subsidiaries, etc.), where the company headquarter has no unconsolidated account, C2: account of a companyheadquarter of a group, aggregating all companies belonging to the group (affiliates, subsidiaries, etc.) where the company headquarter also presents an unconsolidated account, U1: account of a company with no consolidated account, and U2: account of a company with a consolidated account.

³⁴The number of firms with C1 accounts in the final sample is very limited.

A.3 Internal Consistency of Balance Sheet Information

I check the internal consistency of the balance sheet data by comparing the sum of variables belonging to some aggregate to their respective aggregate. I construct the following ratios:

- 1. The sum of tangible fixed assets, intangible fixed assets, and other fixed assets as a ratio of total fixed assets.
- 2. The sum of stocks, debtors, and other current assets as a ratio of total current assets.
- 3. The sum of fixed assets and current assets as a ratio of total assets.
- 4. The sum of capital and other shareholder funds as a ratio of total shareholder funds
- 5. The sum of long term debt and other non-current liabilities as a ratio of total non-current liabilities.
- 6. The sum of loans, creditors, and other current liabilities as a ratio of total current liabilities.
- 7. The sum of non-current liabilities, current liabilities, and shareholder funds as a ratio of the variable that reports the sum of shareholder funds and total liabilities.

After I construct these ratios, I estimate their distribution for each country separately, and exclude the outliers by dropping observations that are below the 0.1 percentile or above the 99.9 percentile of the distribution of ratios. In addition to these filters, I apply further checks:

- 1. In the balance sheet, the sum of the book value of shareholder funds and liabilities (SHFUNDLIAB) and that of total assets (TOTASSTS) should be equal to each other. In the same manner, the book value of shareholder funds (SHFUNDS) should be either equal or less than that of total assets. I drop firm-year observations if they don't satify this accounting rule.
- 2. In the income statement (profit & loss accounts), the monetary value of profitability measures such as netincome (NETINCOME), cash flow (CFLOW), EBIT (EBIT) and EBITDA (EBITDA) should be less than that of either operating revenue (OPRE) or sale (SALE). I drop firm-year observations if they don't satify this accounting rule.

A.4 Further Quality Checks

After I apply the basic cleaning steps listed above, I check the quality of the variables used in the analysis:

- 1. Age: I. construct the variable "age" of the firm as the difference between the year of the balance sheet information and the year of incorporation of the firm (DATEINC) plus one. I drop firms that report dates of incorporation that imply non-positive age values.
- 2. Total Liabilities: As opposed to listed firms, private firms do not report a separate variable "Liabilities." For these firms, there are three ways to construct liabilities:

- (a) Taking the difference between the sum of shareholder funds and liabilities (SHFUNDLIAB) and shareholder funds or equity (SHFUNDS)
- (b) Taking the difference between total assets (TOTASSTS) and shareholder funds or equity (SHFUNDS)
- (c) Taking the sum of current liabilities (CURRENTLIAB) and non-current liabilities (NONCURRLIAB).

I follow first two options to construct "Liabilities." Either gives the same values. I could also have computed liabilities following the third option. However, I find that there are more missing observations if I follow this approach. Nevertheless, for those observations with non-missing information, I compare the values constructed following first two options with those constructed following the last option. I drop firm-year observations where the values are different from each other by 1,500 PPP dollars.

A.5 Winsorization

I winsorize at the 2 and the 98 percentile variables such as logarithm of real sales, logarithm of total assets, collateral, profitability measures, inventory, cash and all debt measures used in the empirical analysis.

Appendix B Composition of Debt Measures

- 1. Current Liabilities (STDebt):
 - Loans (STFinDebt): All short-term financial debts to credit institutions plus part of long term financial debt payable within the year
 - Trade creditors (TC): All debts to suppliers and contractors (accounts payable)
 - Other Liabilities (STOL): Other Short-term debt plus other creditors plus income tax payable, social expenditure payable, dividends payable and other current liabilities
- 2. Non-current Liabilities (LTDebt):
 - Total Long Term Interest Bearing Debt (LTFinDebt):
 - Loans: All long-term financial debts to credit institutions
 - Debentures and Convertible Debt
 - Lease Liabilities
 - Other Long-term Interest Bearing Debt
 - Other Non-current Liabilities (LTOL)
 - Other Long-term Non-Interest Bearing Debt
 - Provisions including pension fund provisions
 - Preferred Taxes
 - Deferred Revenue
 - Minority Interest
- \rightarrow FinDebt=LTFinDebt+STFinDebt
- \rightarrow FinDebtSTOL=LTFinDebt+STFinDebt+STOL
- $\rightarrow FinDebtTOL{=}LTFinDebt{+}STFinDebt{+}STOL{+}LTOL$
- \rightarrow TotDebt=STDebt+LTDebt
- \rightarrow TOL=STOL+LTOL

Appendix C Tables

Country	Pooled	HU	NL	NO	SE	TR
Period	1999-2013	2004-2012	2000-2012	2004-2013	1999-2012	2005-2012
PANEL A: TOTAL ASSETS						
All	40,016	$5,\!532$	191	716	32,895	682
Small	74.8 (29,929)	73.2 (4,050)	2.1(4)	77.4 (554)	76.6 (25,196)	18.3 (125)
Medium	19.9 (7.983)	21.1 (1.168)	16.2 (31)	17.7 (127)	19.3 (6.354)	44.4 (303)
Large	(1,000) 5.3 (2.104)	5.7 (314)	81.7 (156)	(1) (4.9) (35)	(0,001) 4.1 (1.345)	37.2 (254)
Panel B. Employment						
ALL	38.276	4.547	181	579	32,808	161
Micro	66.4	50.8 (2,309)	6.6 (12)	62.7 (363)	69.3 (22,729)	9.9 (16)
SMEs	31.3	(2,303) 44.7 $(2,024)$	44.2	(303) 35.2 (204)	29.1	(10) 57.1 (02)
LARGE	(11,907)	(2,034)	(80) 49.2	(204)	(9,557)	(92) 32.9

Table C.1: PERCENTAGE OF FIRMS IN PERMANENT SAMPLE-BY FIRM TYPE AND COUNTRY

To be Continued.

Table C.2: Percentage of Firms in Permanent Sample-By Firm Type and Count	'RY
---	-----

Country	Pooled	HU	NL	NO	SE	TR
Period	1999-2013	2004-2012	2000-2012	2004-2013	1999-2012	2005-2012
Panel C: Age						
All	39,961	5,506	716	191	32,866	682
Infant	8.5	14.7	1.3	80.1	7.4	3.5
	(3,414)	(811)	(9)	(153)	(2,417)	(24)
Adolescent	(4,759)	(803)	(8)	49.7 (95)	(3,812)	6.0 (41)
Middle-aged	65.2	70.7	11.6	245.0	64.5	60.0
Orb	(20,050)	(3,330)	(00)	(400)	(21,100)	(403)
	(5,752)	(2)	(91)	(0)	(5,451)	(208)

NOTES: Tables C.1–C.2 show the percentages of firms by firm type and country. The firms in each sample refer to ones with non-missing value of the variable on which the percentages are based. In both tables, each cell corresponds to the share of indicated category's number of firms in total economy of the given country-period (%). In Table C.1, shares are constructed based on firm size, and firm size is measured by the logarithm of real total assets and the number of employees, respectively. In Table C.2, firms are categorized based on firm age. In Table C.1, firm size categories are constructed based on predetermined dummies that each equals one if the firm satisfies the criterion for the corresponding firm category at any time during the three years prior to the introduction of LTV ratio cap: SMALL equals one if the given firm's size is below 75^{th} percentile of the distribution, MEDIUM equals one if the given firm's size is above 95^{th} percentile of the distribution. MICRO equals one if the given firm has employees less than 10, SMEs equals one if the given firm has employees between 10 and 249, and LARGE equals one if the given firm has employees higher than 250. In Table C.2, INFANT equals one if the given firm's age is between 3-4, MIDDLE-AGED equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 3-4, MIDDLE-AGED equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 3-4, MIDDLE-AGED equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if the given firm's age is between 5-24, and OLD equals one if t

Period: 1999–2013 Countries: HU, NL, NO, SE, TR					
VARIABLE	Mean	Median	STD. DEV.	25тн Рст.	75тн Рст.
TotDebt	0.57	0.58	0.25	0.38	0.76
FinDebt	0.14	0.0022	0.2	0	0.24
FinDebtTOL	0.46	0.45	0.24	0.28	0.63
STFinDebt	0.023	0	0.054	0	0.0093
STFinDebtSTOL	0.3	0.26	0.19	0.15	0.4
TC	0.11	0.068	0.12	0.017	0.16
Collateral	0.26	0.16	0.26	0.041	0.44
Sales Growth	0.038	0.023	0.35	-0.082	0.15
Profitability	0.15	0.14	0.15	0.063	0.23
Size	13	13	1.7	12	14
Cash	0.21	0.14	0.21	0.032	0.33
Inventory Age	$0.15 \\ 2.7$	$0.05 \\ 2.7$	$\begin{array}{c} 0.2 \\ 0.68 \end{array}$	$\begin{array}{c} 0\\ 2.2 \end{array}$	$\begin{array}{c} 0.24\\ 3.1 \end{array}$

Table C.3: Descriptive Statistics: Permanent Sample

NOTES: Table C.3 reports descriptive statistics of main variables used in the empirical analysis for Full Sample. Debt measures are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit; FinDebt: Total debt excluding trade credit and other liabilities; STFinDebt: Short-term debt from financial institution; STFinDebtSTOL: Short-term Debt excluding trade credit, and TC: Trade Credit. Debt measures are all divided by total assets. Further details on the composition of debt measures are given in the appendix. Collateral is the ratio of total tangible fixed assets to total assets. Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Age is the logarithm of (1+firm age) where firm age in period t that is defined as t minus the date of incorporation plus one. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets.

Sample: Permanent Period: 1999–2013 Countries: HU, NL, NO, SE, TR								
	A	ALL	SM	IALL	ME	DIUM	$\mathbf{L}_{\mathbf{A}}$	ARGE
(% of Total Liabilities)	<u>Mean</u>	<u>Median</u>	Mean	<u>Median</u>	Mean	<u>Median</u>	Mean	<u>Median</u>
STFinDebt	0.034	0	0.027	0	0.052	0	0.056	0
TC	0.11	0.067	0.096	0.052	0.14	0.11	0.12	0.088
STOL	0.52	0.51	0.56	0.56	0.44	0.39	0.44	0.4
LTFinDebt	0.17	0	0.17	0	0.18	0.032	0.13	0
LTOL	0.077	0	0.072	0	0.072	0	0.16	0.061

Table C.4: Composition of Liabilities-By Firm Type

NOTES: Table C.4 reports descriptive statistics of debt measures by different firm size groups. Debt measures are defined as follows. STFinDebt: Short-term debt from financial institutions; TC: Trade Credit, STOL: Other Short-term Liabilities, LTFinDebt: Long Term Interest Bearing Debt, LTOL: Other Long-term Liabilities, and TC: Trade Credit. Further details on the composition of debt measures are given in the appendix

Sample: Permanent Period: 1999–2013 Countries: HU, NL, NO, SE, TR									
	A	All		Small		Medium		LARGE	
(% of Total Assets)	Mean	<u>Median</u>	Mean	<u>Median</u>	Mean	Median	Mean	Median	
Collateral	0.26	0.16	0.26	0.15	0.27	0.18	0.27	0.2	
Inventory	0.15	0.05	0.13	0.02	0.19	0.14	0.17	0.14	
Cash	0.21	0.14	0.24	0.18	0.13	0.07	0.078	0.028	

Table C.5: CASH AND COLLATERAL HOLDINGS-BY FIRM TYPE

NOTES: Table C.5 reports descriptive statistics of cash and collateral holdings by different firm size groups. Collateral is the ratio of total tangible fixed assets to total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Debt measures are all divided by total assets.

SAMPLE: Permanent Period: 1999–2012

Countries: HU, NL, NO, SE, TR

Dependent Variable:	TotDebt	FinDebtTOL (divided by to	TC tal assets)	TC	FinDebtTOL (divideo	TC l by total d	TC ebt)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
$Collateral \times LTV$	-0.009^{***} (-4.915)	-0.013^{***} (-7.760)	0.003^{***} (3.945)	0.003^{***} (3.895)	-0.006^{***} (-4.942)	0.005^{***} (4.250)	0.005^{***} (4.037)
$Cash \times LTV$				-0.0001 (-0.052)			-0.002 (-1.547)
Firm Controls	yes	yes	yes	yes	yes	yes	yes
Number of observations \mathbf{R}^2	$357,622 \\ 0.82$	$357,622 \\ 0.79$	$\begin{array}{c} 348,\!109\\ 0.78\end{array}$	$348,109 \\ 0.78$	$357,622 \\ 0.75$	$\begin{array}{c} 348,109\\ 0.74\end{array}$	$348,109 \\ 0.74$
Firm Fixed-Effects Sector×year Fixed-Effects Country×year Fixed-Effects	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes

NOTES: Table C.6 reports the results of the estimation of equation (4.1). In the estimation, I use Permanent sample that covers the firms from Full sample without non-consecutive yearly observations. The dependent variables are different debt measures i.e., TotDebt, FinDebtTOL, and TC. They are defined as follows. TotDebt: The sum of short-term and long-term debt; FinDebtTOL: Total debt excluding trade credit, and TC: Trade Credit. Debt measures are divided by total assets and total liabilities in Columns 1–4 and Columns 5–7, respectively. Further details on the composition of debt measures are given in the appendix. LTV equals to 1 in the year(s) when LTV ratio cap is in place in the corresponding country. Collateral is a dummy variable that equals one if asset tangibility is higher than its median at any time during three years prior to the application of LTV ratio cap. Firm controls are defined as follows: Profitability is the ratio of EBITDA to total assets. Sales Growth is the logarithmic change of real sales. Size is the logarithm of real total assets. Cash is the ratio of cash and cash equivalents to book value of total assets. Inventory is the ratio of total inventories (raw materials+in progress+finished goods) to total assets. Sectors are classified according to four digit NACE Revision 2 codes. Standard errors are heteroskedastic-consistent errors adjusted for clustering across observations of a given firm, and are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.